

Received 28 December 2022, accepted 16 January 2023, date of publication 31 January 2023, date of current version 8 February 2023. Digital Object Identifier 10.1109/ACCESS.2023.3241146

SURVEY

A Survey on Optical Phase-Change Memory: The Promise and Challenges

AMIN SHAFIEE⁽²⁾, (Graduate Student Member, IEEE),

SUDEEP PASRICHA^{ID}, (Senior Member, IEEE), AND MAHDI NIKDAST^{ID}, (Senior Member, IEEE) Department of Electrical and Computer Engineering, Colorado State University, Fort Collins, CO 80523, USA

Corresponding author: Mahdi Nikdast (mahdi.nikdast@colostate.edu)

This work was supported by the National Science Foundation (NSF) under Grant CCF-1813370, Grant CCF-2006788, and Grant CNS-2046226.

ABSTRACT Silicon photonics (SiPh) technology has facilitated the deployment of integrated photonics across different application domains, from ultra-fast communication in Datacom applications to energy-efficient optical computation in emerging hardware accelerators for machine learning. More recently, the integration of SiPh and phase change materials has created a unique opportunity to realize adaptable, reconfigurable, and programmable photonic platforms. In particular, the nonvolatile programmability in phase change materials has made them a promising candidate for implementing photonic memory cells and architectures. Accordingly, photonic memory systems and even in-memory photonic computing paradigms are on the rise, especially given their potential for improving data access in electronic and photonic processors. However, there are still many challenges in the design and fabrication of phase-change photonic integrated circuits, which need to be addressed. This article presents a comprehensive survey on the recent advances and challenges for the integration of phase change materials with contemporary photonic devices while focusing on the photonic memory application. In particular, we explore phase-change photonic memory from the material level to the architecture level by presenting an overview of different material-level characteristics of phase change materials with their optical, electrical, and thermal properties as well as their integration into SiPh devices and photonic memory architectures and their application for in-memory photonic computing. We also present a comparison with electronic memory and discuss open research challenges that must be addressed to further advance phase-change photonic memory towards successful integration into emerging computing systems.

INDEX TERMS Phase change materials, phase-change memory, in-memory photonic computing, photonic integrated circuits, silicon photonics.

I. INTRODUCTION

With the rapid growth in the diversity and amount of data in emerging data-oriented machine learning (ML) and artificial intelligence (AI) applications, the need for memories with higher cell densities, lower power consumption, higher scalability, longer lifetime, and higher bandwidth has risen. As the size and complexity of emerging applications increase, conventional CMOS-based volatile memories, such as SRAMs and DRAMs, are facing difficulties in meeting energy effi-

The associate editor coordinating the review of this manuscript and approving it for publication was Stanley Cheung^(D).

ciency [1], scalability [2], [3], volume [4], speed, and bandwidth requirements [5]. These shortcomings motivate the consideration of alternative memory technologies in future computing systems, such as phase-change memories. Compared to other nonvolatile memories such as ReRAMs and Flash memories, phase-change memories offer higher stability, enhanced retention time, and higher switching speed [5], [6], [7], [8], [9].

Phase change materials (PCMs) can switch between the amorphous and crystalline states in the presence of an external energy source. Their unique and tunable switching dynamics upon exposure to an external energy source enable them to have two vastly different nonvolatile electrical and optical properties depending on their state. In early 1971, a research group from Bell Telephone Laboratories discovered nonvolatile characteristics of PCMs by using As-Te-I alloys in the very first phase-change-material-based devices, since the material showed two significantly distinct conductivity levels depending on its thermodynamic phase [10]. Later in 1987, many companies such as Intel, IBM, and HITACHI exploited these properties related to phase change materials to manufacture re-writable data storage devices, like CDs and DVDs [10].

Due to nonvolatile phase transitions between the two or more states in PCMs, phase-change memories have recently shown promise in new nonvolatile memory architectures to replace traditional DRAMs [11]. Memory cells based on PCMs may be written to and read from multiple times with substantially reduced area consumption, improved bandwidth, and higher scalability [2], [3]. The energy required to control the state of phase-change memories can be provided either via electrical current (through Joule heating) in electrically driven phase-change memories (EPCMs), or using optical signals (through optical absorption) like a laser pulse in optically driven phase-change memories (OPCMs). In EPCMs, data can be stored by changing the conductivity level induced by the phase transition of the PCM, which in turn depends on the amplitude and duration of the applied electrical current. In contrast, in OPCMs, data is stored in the optical transmission level induced by the phase transition of the PCMs depending on the power and duration of laser pulses. A single OPCM cell can have up to 32 transmission levels, which is enough to store up to five bits per cell [12]. This leads to about two times enhancement in memory performance, about four times lower read energy-per-bit and six times lower write energy-per-bit compared to conventional DRAMs [5], [12].

In addition to having more energy-efficient data storage, reducing energy and latency for data movement on electrical links has always been a challenging task in CMOS integrated circuits. For instance, on a modern CMOS computing system, 0.1-0.2 pJ/bit is required to transfer data over a 1-mm-long electrical interconnect used for on-chip communication, such as in the case of a processing core accessing an L2 cache bank. This number goes up to 1-4 pJ/bit over longer electrical links, such as in the case when a core accesses remote L3 cache banks. In addition, it takes up to 30 pJ/bit for off-chip communication when a core needs to access the CMOS main memory (i.e., DRAMs) [13], [14], [15]. While the absolute values of these energy costs may seem low, in fact, they are significantly higher than the energy budget required for future computing systems. This data movement overhead imposes severe limitations on the scalability of CMOS integrated circuits [13]. To address the aforementioned limitations, due to their compatibility with CMOS integrated technology, silicon photonic (SiPh) devices have been widely employed in emerging integrated circuits to exploit photons instead of electrons to transfer data [16]. In particular, state-of-theart SiPh links have been used to support on-chip and offchip communication with substantially lower energy consumption and latency over longer distances, with a higher bandwidth (e.g., ≈ 51.2 Tb/s compared to 112 Gb/s for stateof-the-art electrical links) [17]) than conventional electrical links. Thus, SiPh offers scalability with enhanced energy efficiency and performance and compatibility with contemporary CMOS fabrication technology, making it an attractive choice to support data movement in future processing chips. For example, the work in [5] suggests using SiPh links instead of electrical links to facilitate the data movement to and from OPCM cells to further minimize the latency and static power consumption in future photonic memory systems [5].

Despite the benefits that OPCMs may offer, research suggests that implementation of OPCM is challenging due to its essential need for electro-optical and opto-electrical conversions to convert the electrical signals coming from a memory controller to optical signals and vice versa. Consequently, one cannot simply replace a DRAM cell with an OPCM cell. Therefore, the entire memory architecture needs to be redesigned to make it compatible with OPCM arrays. In addition, OPCM cells are extremely sensitive to thermal variations stemming from the static power consumption of nearby photonic and electronic components. This can lead to variations in optical transmission level, hence increasing the write and read error rates [5]. A higher area consumption compared to conventional DRAM main memories can be considered as another challenge facing the implementation of OPCMs.

In this paper, we present a comprehensive survey on recent advances and challenges in the application of PCMs from material-level to system-level in photonic memory and in-memory computing systems. In addition, we discuss simulation methods and challenges facing the design of phasechange memories and in-memory computing units based on PCMs in detail. A few review papers on similar topics have been published in recent years. The work in [18] presents a survey on the application of PCMs for reconfigurable silicon photonics. Starting from a brief overview of the optical properties of PCMs, this work focuses on the application of PCMs in silicon-photonic switching networks and modulators. In addition, it presents a brief survey of the application of PCMs in photonic computing units. However, the paper's outlook and scope are limited to device-level designs. Hence, the challenges of PCM-based memory and in-memory computing systems have not been discussed. The work in [19] and [20] also presents a survey on photonic devices based on PCMs. However, these papers focus on the thermodynamic and switching properties of PCMs and their application in reconfigurable photonic devices, such as active plasmonics, metamaterials, and color displays. They do not discuss PCM-based photonic memory and in-memory computing systems and related open challenges, as it is done in this survey. The organization of this paper is as follows:



FIGURE 1. (a) Set and reset procedures in phase-change memories. (b) The most conventional Ge-Te-Sb alloys for nonvolatile photonic memory applications [18]. Permission: https://creativecommons.org/licenses/by/4.0/.

- Section II presents an overview and fundamentals of PCMs and their working mechanisms for memory applications, along with their thermal, electrical, and optical properties as well as their physical and thermodynamic properties.
- Section III provides an overview of the applications of PCMs in data storage cells for photonic memories and their different design approaches and read and write policies. Moreover, we review recent phasechange memory architectures and applications of PCMs to implement in-memory computing units for computationally expensive applications, such as ML and AI applications. Furthermore, this section describes the simulation of a single photonic memory cell based on GeSbTe alloys (GST) for photonic memory applications.
- Section IV presents a bottom-up overview of the design challenges of OPCMs and their architectures. Challenges such as power and latency trade-offs, the complexity of OPCM's designs, and endurance will be discussed in this section. Some of the state-of-the-art applications of phase change materials for programmable photonics and photonic tensor cores will be also discussed in this section.
- Section V concludes the paper and presents a summary of different applications of PCMs reviewed in this paper.

II. PHASE CHANGE MATERIALS: AN OVERVIEW

In this section, we present an overview of PCMs, including their thermal, optical, and electrical properties.

A. FUNDAMENTALS OF PHASE CHANGE MATERIALS

The molecular structure of a material can have a big impact on its optical and electrical characteristics. Furthermore, even the same material with a specific molecular structure at room temperature, such as silicon (Si), can have variable optical and electrical characteristics depending on its crystallographic orientation. When heated with an external energy source (i.e. electric field or optical signal), the temperature of PCMs starts to increase by absorbing the energy of the heat source. One important material parameter related to PCMs is the melting temperature (T_l) . The regions of the material after exposure to the heat source that has a temperature above T_l will be melted and quenches (cools down with the rates higher than 10^9 K.S⁻¹ [21]) in about 700 picoseconds [22]. The quenched region will have an amorphous structure regardless of the initial state of the material. This process is called reset (see Fig. 1(a)) or amorphization. The temperature distribution of the material with a given geometry can be obtained by solving the unsteady-transient heat flow equation [22]. The simulation approaches of PCM-based photonic devices will be explained in detail in Section III-E.

PCMs can repeatedly change their phase from amorphous (crystalline) to crystalline (amorphous) states (see Fig.1(a)). In addition, PCMs can be fully crystallized, fully amorphous, or partially crystallized/amorphous (intermediate state) depending on the duration and power of the heat source. Hence, different electrical and optical properties at each state can be achieved. Typically, we desire to have a low T_l because it will result in low power and fast reset (amorphization) of the material. Another important material parameter is the crystallization temperature (T_g) . When the material in the intermediate state absorbs energy through an energy source, the regions of the material which have higher temperatures than T_g and lower than T_l will recapture the crystalline structure. This process is called set (see Fig. 1(a)) and can take hundreds of nanoseconds or higher [21]. High T_g is desirable to improve the phase stability of the material, which is crucial to enable reliable data retention in nonvolatile memories [23].

The response time of nonvolatile memory cells plays an important role in determining the average memory-access latency in memory systems. The average memory-access latency needs to be as low as possible. The reason for this is that in highly parallel processors, most of the time is consumed during the multiple access to the main memory per cycle, and having high average memory-access latency can act as a bottleneck for the speed of the whole system. Thus, the lower average memory-access latency leads to a higher speed of the whole processing system [24], [25]. The average memory latency can vary from hundreds of picoseconds to hundreds of nanoseconds depending on the architecture of the cell types and the memory architecture. High read and write latency of the memory cells lead to higher average memory latency and lower read and write throughput of the memory systems.

The average memory latency can be decreased by increasing the number of bits stored in a single OPCM cell [5]. The very first materials that showed such a fast response, as well as high optical contrast between the two phases to

store logic 1 or logic 0, were GeTe and GeTeSnAu alloys [26]. In addition, pseudo-binary alloys of GeTe-Sb₂Te₃ like Ge₁Sb₄Te₇, Ge₁Sb₂Te₄, and Ge₂Sb₂Te₅ (known as GST) were used for photonic memory applications (see Fig. 1(b)) [26]. The state-of-the-art OPCMs require 25 nanoseconds to be written and read [25]. In addition, PCMs must have a high optical transmission contrast (when the optical transmission level is being used to store the data) and a high resistance contrast for different states (when conductivity or resistance level is being used to store the data), long durability over time, and a large capacity to be considered as a suitable candidate for data storage applications. Among the aforementioned materials, GST is the most popular material for photonic nonvolatile data storage because of having distinctive optical transmission and resistance levels related to crystalline and amorphous states. Conventional PCMs (e.g., GST) have a fast transition speed and different distinctive optical and electrical properties depending on their phase. PCMs tend to maintain their state even when the external energy source is turned off, making them a promising alternative for developing nonvolatile photonic memory cells due to their dynamic and adjustable optical and electrical characteristics in each state [10], [26], [27], [28], [29], [30].

B. THERMAL PROPERTIES

One of the most important parameters related to PCMs is thermal conductivity which determines the rate of heat transfer in the material. As the thermal conductivity increases, the material is able to absorb and transfer the heat faster. In the previous section, we discussed how the energy of a heat source will be absorbed by the material where this energy will then trigger the phase transition of the PCM. PCMs have a relatively low thermal conductivity which means that they are unable to quickly absorb the energy from a material with low thermal conductivity. This makes the phase transition to be slow in PCMs. As we enhance the thermal conductivity of the material, lower energy will be needed to induce the phase transition in the material because a higher portion of the absorbed energy contributes to the temperature variation of the material [31]. The work in [32] and [33] reports experimental results on thermal conductivity for the most conventional PCMs (GST) in amorphous and crystalline states. We can see from Table 1 that for GST, the thermal conductivity for amorphous and crystalline GST is about 0.19 and 0.57 W/mK, respectively. This is relatively low compared to conventional semiconductor materials such as silicon with thermal conductivity of 148 W/mK at room temperature. Table 1 shows the thermal properties of different materials that can be used to obtain a single photonic memory cell for data storage.

Another important thermal property related to PCMs is the thermo-optic effect. So far, we have discussed how phase transitions in PCMs can be induced by optical or electrical signals. However, considering the physics of the devices, any change in the temperature of the material can impose a change in its optical properties, such as the material's refractive index and extinction coefficient via thermo-optic effect. The refractive index is a parameter that determines the group velocity of an electromagnetic wave in a medium. Also, the extinction coefficient is a parameter which describes the loss (absorption) of an electromagnetic wave in a medium. The thermo-optic coefficient of a material quantifies the changes in its optical properties in the presence of thermal variations. The work in [36] proposed analytical models to monitor the thermo-optic effect and changes in the refractive index and extinction coefficient in GST caused by temperature variations. The work in [36] suggests that the change in the refractive index (n) of PCMs leads to an excessive phase shift while the change in the imaginary part (κ) leads to additional attenuation caused by material loss. Therefore the ratio of the change in the real part of the refractive index to the change in the imaginary part of the refractive index $(\Delta n / \Delta k)$ was used as a figure of merit for the proposed analytical models in [36]. They captured the real and the imaginary part of the thermo-optic coefficient of GST experimentally and showed that the real and imaginary parts of the crystalline and amorphous GST can change significantly under temperature variations.

C. ELECTRICAL PROPERTIES

The current flow in PCMs has a different nature in comparison to semiconductors. Because of their molecular structure, resistance in PCMs can be described by electrons hopping between traps in the PCM's lattice. Like any other molecular structure, the current in a PCM can be modelled via Poole-Frenkel transport of electrons through traps [10], [26]. Therefore, the current flow in a PCM depends on the distance and the potential barrier among traps. Due to the crystallographic structure of PCMs in amorphous state, the potential barrier between traps is higher. Thus, PCMs in the amorphous state have much higher resistance than in the crystalline state. By applying an electric field to a PCM, the potential barrier between traps can be lowered, and so will the resistivity of the PCM in the amorphous state. Experimental results show that such resistivity decreases linearly with an increase in the electric field [10]. Further increasing the applied electric field will lead to a collapse of the local electric field between traps, which leads to a negative resistance in the material and a sharp drop in the resistivity without any phase change. In a negative resistance state, switching off the voltage source will lead to regaining the original resistance level of the material in amorphous state in the absence of the electric field. Applying a sufficiently high electric field ($\approx 562 \text{ MV/m} [37]$) over a specific time period (≈ 100 ns [37]) will lead to reaching crystallization temperature (T_c) and a phase change from amorphous to crystalline state.

A PCM in the crystalline state has higher conductivity compared to amorphous state. This property is nonvolatile and revertible, meaning that switching off the voltage source will not change the resistivity of the PCM in the crystalline

Material	Thermal Conductivity (W/mK)	Heat Capacity (C_p)—(J/gK)	Density (ρ)—(gr/cm^3)
a-GST [33]	0.19	0.213	5.87
c-GST [33]	0.57	0.199	6.27
a-GSST [34]	0.2 ± 0.02	1.5 ± 0.1	5.27
c-GSST [34]	0.48 ± 0.06	1.8 ± 0.1	5.53
Si ₃ N ₄ [35]	10-43	0.673 - 1.100	2.37-3.25

TABLE 1. Thermal properties of common phase change materials used in phase-change memory cells (a: amorphous and c: crystalline).

state. The same procedure can be exploited to change a PCM's state from amorphous to crystalline [10], [26], [27], [28], [29], [30]. Depending on the electrical current's amplitude and duration, energy will be provided for a phase-change memory cell through joule's heating (Ohmic heating) to trigger the phase transition to perform write or reset.

D. OPTICAL PROPERTIES

Understanding the optical properties in PCMs is of the essence when it comes to their application in photonic integrated circuits. Depending on their state, PCMs have various optical characteristics. As a general rule and in comparison to the amorphous state, PCMs in the crystalline state are extremely absorbing due to their high extinction coefficient and this is the main reason for the optical transmission contrast they exhibit. This property makes PCMs suitable for implementing phase-change memories by storing logical bits and decoding them into the optical transmission levels of the material originating from phase transitions. PCMs can also take intermediate states, which means that some portion of them can be in the crystalline state while the remaining portion is in the amorphous state.

PCM's effective permittivity determines the electric polarizability. Electric polarizability describes the dipole moment of the material when subjected to an electric field and directly affects the material's dielectric constant and refractive index [40]. The effective permittivity (ε_{eff}) of a PCM in an intermediate state, can be estimated using the Lorenz mathematical model [22], [41]:

$$\frac{\varepsilon_{eff}(\lambda) - 1}{\varepsilon_{eff}(\lambda) + 2} = X_f \times \frac{\varepsilon_c(\lambda) - 1}{\varepsilon_c(\lambda) + 2} + (1 - X_f) \times \frac{\varepsilon_a(\lambda) - 1}{\varepsilon_a(\lambda) + 2}.$$
(1)

Here, X_f is the crystalline fraction and takes a number between 0 and 1, illustrating the portion of the PCM which is in the crystalline state. Moreover, the wavelength-dependent dielectric permittivity function ($\varepsilon(\lambda)$) can be calculated as:

$$\varepsilon_a = n_a^2,\tag{2}$$

$$\varepsilon_c = n_c^2, \tag{3}$$

where n_c and n_a are the complex refractive indices of the PCM. Finally, using (1), the real and the imaginary part of the effective refractive index—which determines the phase delay and absorption of the light in a material—of a PCM in

an intermediate (mixed) state can be estimated as [22]:

$$n_{eff} = \sqrt{\frac{\sqrt{(\varepsilon_1 + \varepsilon_2)^2} + \varepsilon_1}{2}},\tag{4}$$

$$k_{eff} = \sqrt{\frac{\sqrt{(\varepsilon_1 + \varepsilon_2)^2} - \varepsilon_1}{2}}.$$
 (5)

In (4) and (5), ε_2 and ε_1 are the real and imaginary part of $\varepsilon_{eff}(\lambda)$ in (1) [22].

The refractive index (*n*) and extinction coefficient of conventional (κ) PCMs are depicted in Fig. 2. We can observe that the materials in the crystalline state have a much higher extinction coefficient compared to their amorphous state. This means that materials in the crystalline state absorb more optical power compared to the amorphous state and convert it into heat to be used to trigger the phase transition. The amount of absorbed optical power in the crystalline state because of the imaginary part of the refractive index in the crystalline state is significantly higher compared to the amorphous state. The absorption coefficient— α (cm^{-1})—of PCMs can be calculated from their extinction coefficient (κ) for any given wavelength according to the following model [10]:

$$\alpha = \frac{4\pi\kappa}{\lambda}.\tag{6}$$

Note that when the extinction coefficient is higher, the absorption coefficient of the material is also higher, and hence laser pulses with lower power and duration are needed to trigger the phase transition. From Fig. 2(c), we can observe that GST has reasonably high absorption due to its high extinction coefficient in the crystalline and amorphous state for O-band (1260 nm–1360 nm) and C-band (1530 nm–1565 nm), compared to other PCMs which make the phase transitions efficient in terms of latency and energy consumption. It is also shown in Fig. 2 that the extinction coefficient in the amorphous state is lower than that in the crystalline state which, compared to the crystalline state, results in more transparency of PCMs in the amorphous state as well as lower absorption in this state.

III. PHOTONIC MEMORY BASED ON PHASE CHANGE MATERIALS

In this section, we review the state-of-the-art photonic memory storage cells as well as their advantages and challenges, recent work on photonic memory architectures, and finally, some recent efforts on using PCMs for photonic in-memory computing.



FIGURE 2. Refractive index profile of the most conventional phase change materials. *n* is the real part of the refractive index and *k* is the imaginary part of the refractive index (extinction coefficient) [38], [39]. Permission: https://creativecommons.org/licenses/by/4.0/.

A. EPCMs AND OPCMs

In Section II-A, we showed that to trigger phase transitions in a PCM, the material should absorb energy, which leads to temperature variation in the PCM. The required energy for a phase change can be provided by absorbing the energy from an external heat source with a specific duration and power level. For example, the required energy can be provided via electrical current or optical laser pulses. Because of light absorption in PCMs, optical signals can be used to trigger the phase transition and optically control PCMs.

In electrically controlled PCMs [42], [43], [44], electrical current or bias voltage can be used to trigger the phase transition in the PCMs. By applying an electric field, the joule heating in the PCM can trigger the phase transition. Depending on the amorphous/crystallized portion of the material, the resistance and optical transmission in the PCM will change. Eventually, the state of the PCM can be read via an optical or electrical read signal. For hybrid cells, it is possible to set the state of the phase-change memory cells using an optical signal and read the state using an electrical signal, and vice-versa [45], to reduce the energy consumption of the cells. To take advantage of the interesting properties of PCMs to implement phase-change memories, one should have a clear insight into their electrical and optical properties.

The most conventional way to exploit a PCM for memory architectures is to control the cells with an electrical field, known as electrically-driven phase-change memories (EPCMs). The work in [46] proposed a method to architect EPCMs to address the scalability issue of conventional DRAMs. Their baseline design showed higher energy consumption and latency compared to conventional DRAMs. However, they realized that using EPCMs instead of conventional DRAMs could help further scale the architecture. There are several other memory architecture designs based on EPCMs [6], [47], [48]. Table 2 lists some advantages and disadvantages of EPCM-based architectures compared to conventional DRAMs based on some recent work in this area. Compared to DRAMs, EPCMs can store a higher number of bits per cell (up to 2 bits per cell [5]), leading to multilevel cells (MLCs), and they are more scalable and stable under thermal variations. However, they suffer from higher latency

TABLE 2. Advantages and disadvantages of EPCMs compared to DRAMs.

Advantages	Disadvantages
Multilevel cell (MLC) [6], [9], [48]	Higher latency [46]
Scalability [46]	Higher energy consumption [6], [9], [48]
Thermal stability [50]	Charge pump circuit requirement [51]
—	Lower endurance [52]

and energy consumption (e.g., ≈ 50 ns for reset and ≈ 60 ns for read [5]), require additional circuitry such as charge pump circuits, and have lower endurance because of aging of EPCM cells (e.g., 1e8 writes compared to DRAMs that support >1e16 writes [6], [9], [46], [48], [49].

Despite several advantages of EPCMs, such as enhanced MLC capacity with up to 2 bits per cell [5], [6], they suffer from high power consumption and latency. This limits the employment of EPCMs in emerging nonvolatile data storage architectures. To overcome these limitations, optically driven phase-change memories (OPCMs) have been recently introduced that use electromagnetic waves instead of electrons to control the state of phase-change memory cells. OPCMs will be discussed in detail in Section III-B. Here, we give a comparison of OPCMs against EPCMs. Most of the limitations related to EPCMs can be addressed by exploiting OPCMs in the optical domain. Table 3 highlights the comparison between OPCMs and EPCMs. Compared to EPCMs, OPCMs offer lower latency and power consumption (e.g., 60 to 130 pJ with 25 ns for read and write [12]), higher MLC capacity (up to 5 bits per cell [12]), and higher bandwidth (e.g., 1 GHz [53]) with a lifetime of 1e6–1e8 writes per cell [54]. However, such benefits come at the cost of using a complex electro-optic unit, re-design of read and write policies, accumulated optical loss and crosstalk upon scaling, need for silicon photonic links, and sensitivity to thermal variations. Due to the novel nature of OPCMS, the research related to exploiting OPCMs for scalable nonvolatile photonic memory architectures is constantly evolving [5], [12], [18], [36], [53], [55].

B. OPCM STORAGE CELLS

In this section, we first discuss the compatibility of silicon nitride and silicon platforms with PCMs as well as the benefits each platform offer when integrating with PCMs.



FIGURE 3. (a) Schematic of a photonic-memory cell using Si₃N₄ waveguide and GST. Write and erase can be performed by applying optical signals. (b) Transmission change for the amorphous and crystalline state. [57], [58]. Permission: https://creativecommons.org/licenses/by/4.0/.

TABLE 3. Advantages and disadvantages of OPCMs compared to EPCMs.

Advantages	Disadvantages
Low latency [18], [36], [55]	Need for electro-optic units [5]
Low energy consumption [55]	New read and write policies [5]
Higher MLC capacity [56]	Need for silicon photonic links [5]
Higher bandwidth [53]	Optical loss and crosstalk [13]
—	Sensitive to thermal variations [5], [36]

Moreover, we present a comprehensive review of different designs to implement single-bit and multi-bit OPCM cells and a comprehensive review of plasmonics-based OPCM cell's performance.

When it comes to integrating PCMs with state-of-the-art photonic integrated devices and circuits, the substrate material is very important as it will directly impact the system performance in terms of speed, energy efficiency, and footprint. Over the past few years, Si₃N₄ has been the preferred substrate to be integrated with PCMs due to its lower thermal conductivity (\approx 43 W.m⁻¹.K⁻¹ [59]) compared to silicon $(\approx 148 \text{ W.m}^{-1}.\text{K}^{-1}$ [59]). Accordingly, the minimum energy required to trigger a nonvolatile amorphization is lower when using Si₃N₄ (e.g., 42 pJ) compared to silicon-based GST (e.g., 388.4 pJ) [59]. This makes the Si_3N_4 platform much more energy efficient compared to silicon platforms for PCM integration. However, silicon could offer other advantages over Si₃N₄ such as a smaller footprint, enhanced mode confinement, higher speed, and better integration with CMOS integrated circuits.

The work in [59] presented a detailed experimental analysis of the performance of silicon and silicon-nitride-based OPCMs. It was shown that integrating GST on top of a Si₃N4 waveguide leads to a propagation loss of 0.079– 2.470 dB/ μ m, which is higher than silicon-based GST cells at 0.059–1.445 dB/ μ m in the C-band. The reason for the lower propagation loss in silicon-based OPCMs can be explained via refractive index contrast between silicon and GST. The refractive index contrast between silicon nitride and GST. This leads to lower scattering loss and reflections, and ered as upper cladding in [59]) is another reason for the lower propagation loss of silicon-based OPCMs compared to those based on silicon nitride. The large contrast in the refractive index of silicon and air leads to better confinement of optical modes in silicon-based waveguide. Therefore, waveguides with smaller widths (e.g., 500 nm compared to 1.3 μ m for silicon nitride [59]) will be needed so that the confined light in the waveguide can interact with GST. This reduces the evanescent coupling of the light to the GST, resulting in a lower propagation loss in silicon-based OPCMs compared to those based on silicon nitride [59]. In [59], it was shown that using silicon platforms can lead to a more compact footprint, higher energy consumption, and reduced amorphization time. Silicon compatibility with

hence better matching of the optical mode in the underlying waveguide and GST. Therefore, a lower propagation loss is observed. The large refractive index contrast between silicon and the upper cladding of the waveguide (the air was consid-

to a more compact footprint, higher energy consumption, and reduced amorphization time. Silicon compatibility with CMOS foundries makes it the preferred platform for Telecom, Datacom, data processing, and in-memory computing applications. On the other hand, silicon nitride is the preferred platform for photonic memories due to its lower thermal conductivity and hence higher energy efficiency. Higher thermal conductivity of silicon results in higher power dissipation to achieve the needed transmission contrast due to the presence of PCM, and this leads to the lower energy efficiency of silicon platforms compared to silicon nitride.

State-of-the-art reconfigurable SiPh integrated circuits operate based on the thermo-optic or free-carrier-dispersion effect of silicon. Such devices suffer from multiple limitations such as a large footprint (e.g., >100 μ m) and high power consumption (e.g., >1 mW) [60]. This is one of the main motivations for using PCMs in SiPh integrated circuits. Regarding the choice of PCMs, among the different options we introduced in the previous section, GST is the most popular one to be integrated with silicon-on-insulator (SOI) and silicon nitride photonic devices. The reason for this is the high optical contrast between two amorphous and crystalline states in GST and the stability of crystallographic phases over time compared to other options [60]. Apart from GST,



FIGURE 4. Different designs for photonic memory storage cells: a) Using partially etched MRRs. (b),(c) Simulated optical modes using COMSOL Multiphysics for a waveguide with GST on top in amorphous and crystalline state, respectively. (d),(e),(f) Fabricated storage cells with MRRs, MZIs, and beam splitters, respectively [62]. Permission: https://creativecommons.org/licenses/by/4.0/.

Ge-Sb-Se-Te alloys (GSST) are another type of PCMs with great promise to be integrated with photonic devices. GSST offers enhanced thermal conductivity (see Table 1) and lower optical power loss compared to GST. Optical loss is a parameter that lower the optical signal amplitude from the input to the output of a photonic device. Optical power loss is one of the main factors that limits the scalability of silicon photonic integrated circuits. GSST also shows better lifetime and thermal stability compared to GST [61].

Considering integrating GST with silicon nitride platforms to implement OPCM cells, the work in [53] proposed a design based on integrating a GST thin film with a length of 5 μ m and a thickness of 10 nm with a silicon nitride ridge waveguide (see Fig. 3). This design is able to store a logic bit "0" or "1" depending on the state of the GST on top of the Si₃N₄ ridge waveguide. When the amorphous GST turned into the crystalline state upon exposure to an optical signal, the transmission of the GST changes by about 21%, which corresponds to logic bit "0" (erase procedure). This is achieved by applying intense laser pulses for a duration of 100 ns with a total energy of 533 pJ. Since the crystalline GST absorbs almost 80% of the input power, 430 pJ energy is needed to trigger the phase transition from crystalline to amorphous state. The reason for a lower energy requirement to change from the crystalline to amorphous state is the higher absorption coefficient (or higher extinction coefficient as shown in Figs. 2(c) and 2(d)) of GST in the crystalline state compared to that in the amorphous state. It was also demonstrated in [53] that the optical transmission contrast can be increased by increasing the length of the GST. The PCM-based photonic memory cell that was proposed in [53] is able to store only a single bit (either a "0" or a "1"). This puts additional limitations on the scalability and power consumption of photonic memory arrays build using such a cell to store more than one bit per cell based on this design. Leveraging the single storage cell proposed in [53], multiple optical storage cell structures can be designed using passive silicon photonic devices, such as those based on beam splitters, Mach–Zehnder interferometers (MZIs), and microring resonators (MRRs). Some of these example designs for a single optical storage cell are depicted in Fig. 4.

The work in [62] performed a detailed analysis of the performance of different photonic cells demonstrated in Fig. 4 with different geometries for broadband photonic applications. In this work, performance optimization of the cells was performed by altering the PCM cell geometry. It was shown for the race-track MRRs in Fig. 4(a) that different optical parameters, such as quality (Q)-factor, extinction ratio, and resonant wavelength, can be used to read the GST state because of the significant contrast between their values for the amorphous and crystalline state. For instance, it was shown that there can exist a significant difference in the O-factor when the phase of the PCM (GST was used in [62]) changes from the amorphous to crystalline state. In addition, it was shown that the contrast between Q-factor of a single cell increases as the width of the GST cell increases. Using a single-cell OPCM, the work in [63] showcased how a single MRR-based OPCM cell can be scaled to 256 cells by cascading them to obtain an OPCM-based architecture with 512-bit capacity. The proposed architecture was also experimentally tested to store a small digital image. In this design, each OPCM cell is able to store only two bits. An optical pulse was used with a total energy of 890 pJ for amorphization of the GST in each cell, which makes this design extremely power-hungry as well as having high write latency. In addition, using this design in a photonic memory architecture leads to constraints on parallelism because of the lower MLC capacity (i.e., two bits per cell).

PCMs like GST show better optical transmission contrast in certain wavelength bands. This can be observed by considering the refractive index and extinction coefficient in PCMs depicted for different wavelengths in Fig. 2. As a result, having an OPCM cell design that can operate in a broad wavelength range is important. This is also helpful because of its effect on the scalability and energy efficiency of the OPCMs. Having OPCMs working with a broader wavelength range enables using a larger number of bit-lines to address, read, and write cells [5]. But realizing OPCMs working with a broader wavelength range is challenging because of PCMs' refractive index profile. Taking GST as an example, from Fig. 2(c), we can see that the change in the refractive index and extinction coefficient is significant under a wide wavelength range (e.g., 800–1600 nm in Fig. 2(c)). This will prevent having stable transmission levels over a wide range of wavelengths and can lead to a reduction in OPCMs' MLC capacity and variation in the transmission level of the cell over time.

The work in [61] showed that by using a PCM based on GSST ($Ge_2Sb_2Se_4Te_1$) one can achieve a 2-bit nonvolatile photonic storage cell to work with a broader wavelength range (1–18.5 μ m) and with enhanced crystallization ability. GSST can be also exploited to tackle problems related to the variations in the optical properties of GST caused by thermal



FIGURE 5. (a) Fabricated photonic memory storage cell. (b) Programming and erasing scheme of the cell in C-band and L-band. (c) Change in crystalline area of the cell and therefore transmission against the pulse energy. (d) Finite element modelling (FEM) simulation of the sample to solve the heat transfer equations in GST based on the input energy of the pulse. The dark regions have amorphous state [12]. Permission: https://creativecommons.org/licenses/by/4.0/.

variations in photonic applications [36] because of having a lower thermo-optic coefficient and being less susceptible to thermal variations, which results in having higher thermal stability. In [64], it was shown how using GSST instead of GST in MZI- and directional couplers (DC)-based photonic switches can enhance the insertion loss and crosstalk noise. For a single switching element using GSST, an insertion loss of 0.04–0.4 dB and crosstalk of -32 dB were obtained [64].

The attenuation coefficient is a parameter that can be used to analyze the switching dynamics of PCMs. The attenuation coefficient can be defined as the amount of optical power being absorbed by a PCM per unit length. The work in [65] developed analytical models to model the attenuation coefficient due to absorption of GST in the amorphous and crystalline state. Having such a model helps analyze and optimize the switching dynamics of GST for photonic switching networks and data storage applications. Leveraging such models, [65] aimed at optimizing different parameters such as thickness, width, and length of GST cells for fast and energy-efficient multilevel switching of optical data storage units and photonic switching networks. It was shown that as the width of the GST sample increases, the absorbed power in GST will increase and the input optical power will be attenuated evanescently along the length of GST.

As discussed before, PCMs can take amorphous state, crystallised state, or amorphous/crystalline (intermediate) state. Recall that by controlling the crystallization fraction of a PCM used in a phase-change memory cell, we can realize multilevel photonic memory data storage cells [11], [12]. Such a behavior is shown in Fig. 5 where the crystalline fraction of the PCM can be controlled through the input optical signal energy. From Fig. 5, we can see that for the given $2-\mu$ m-long GST sample, 180 pJ is needed to induce a 60% change in the crystallization fraction. Note that the initial state of the GST sample in this work was the crystalline

state. The crystallization fraction of PCMs can be controlled by pulse amplitude modulation (PAM) or pulse width modulation (PWM).

The work in [12] experimentally realized up to a 5-bit OPCM cell by controlling the crystallization fraction which leads to obtaining up to 32 different transmission levels. A 50-ns laser pulse was used with variable amplitude (depending on the number of the bits) to program the cell as well as a 50-ns high amplitude laser pulse followed by a 200-ns low amplitude laser pulse to erase the cell. We can observe from Fig. 5 that programming the cell—which implies the phase transition of crystalline to amorphous—takes multiple pulses with the same duration but different power. In addition, [12] proposed a novel double-step pulse controlling for phasechange memory cells as well as a detailed analysis of pulse parameters on the multilevel cell dynamics.

Plasmonic devices are a category of photonic devices which deal with the interaction of free electrons (metals) and an electromagnetic wave, like light. Upon interaction with incident light, the free electrons in the metals start oscillating. The oscillation of electrons, which is called the plasmonic effect, can be modeled via the Drude model and it is called surface plasmon [66]. The latency and energy efficiency of a single phase-change memory cell can even be further enhanced by improving the PCM-based photonic memory cell's design. The work in [67] presented a novel method to enhance the switching capability of a GST cell in terms of latency and energy efficiency for photonic memories. In particular, it was shown that integrating the GST cell with two sub-micron nanoantennas (made with silver) can significantly enhance the light-matter interaction between SiN waveguide and GST through the plasmonic effect. Therefore, this leads to write/erase speed of 2-20 ns and write/erase energies of 2-15 pJ, which is about 2 orders of magnitude improved compared to contemporary multilevel GST cells for photonic memories.

As another example of exploiting the plasmonic effect to enhance the light-matter interaction in OPCM cells, the work in [68] proposed an all-optical OPCM cell using the plasmonic effect. It was shown that by integrating silver nanostructures with GST, the performance of an MRR-based OPCM cell in terms of insertion loss, optical transmission contrast, and area consumption can be significantly improved. Thus, higher number of bits can be stored in a single OPCM cell with lower area consumption and latency, which makes the memory more scalable and energy efficient. The schematic of the design is shown in Fig. 6. As it can be seen from the figure, the design consists of an MRR-based OPCM in which the GST is surrounded with silver nanostructures that lead to enhancement in the GST-light interaction through the plasmonic effect. Three different designs based on silver/GST fabrication were investigated. In the first design, the silver/GST structure was fabricated on top of the silicon-nitride waveguide and is called the non-embedded design (see Fig. 6 (b)). The second design includes fabricating the silver/GST structure halfway inside the silicon-nitride



FIGURE 6. (a) Plasmonically enhanced all-optical MRR-based OPCM. (b) The silver/GST structure fabricated above a silicon-nitride waveguide. (c) The silver/GST structure fabricated halfway in a silicon-nitride waveguide. (d) The silver/GST structure fabricated in a silicon-nitride waveguide [68]. Permission: https://www.mdpi.com/openaccess.

waveguide and is called the half-embedded design (see Fig. 6 (c)). In the last design, the silver/GST structure was fabricated inside the silicon-nitride waveguide and is called the full-embedded design (see Fig. 6 (d)). The experiments in [68] showed 13.7% optical contrast between the amorphous and crystalline state when the silver/GST structure is fabricated above the silicon-nitride waveguide (the nonembedded design). This was significantly higher than the other two cases: 9.6% for the half-embedded and 6.1% for the full-embedded design. In addition, for the non-embedded case, a maximum insertion loss of 2.3 dB and a minimum insertion loss of 1 dB per cell was reported for the GST in the amorphous and crystalline state, respectively. The overall insertion loss was higher for the other two design cases. Finally, [68] showed that their design can gain from a more compact footprint per OPCM cell with dimensions of 50 \times 90 nm², which is significantly smaller compared to traditional OPCM cells.

C. OPCM ARCHITECTURES

As the size and complexity of data-driven applications increase, memories with a higher cell density, lower power consumption, and higher bandwidth are required to carry out computationally intensive operations. The work in [69] focused on simply replacing the electrical links with silicon photonic links without redesigning the memory architecture to enhance the latency of EPCMs. The approach requires multiple electro-optical (E-O) and opto-electrical (O-E) conversions to convert the electrical signals to optical signals and vice versa. As the number of O-E and E-O conversions increases, the energy efficiency in the memory architecture will decrease. Note that EPCM cells are being used in memory architectures because of their benefits in terms of scalability and thermal stability [46].

OPCMs can alleviate the scalability, power consumption, and bandwidth limitations of conventional memories, such as DRAMs. However, it is inefficient to simply exchange the EPCMs with OPCMs in the same architecture because the electrical signals coming from the memory controller cannot be directly employed to write, read, or program the OPCMs.

11790

Therefore, to adapt the OPCMs to state-of-the-art memory architectures, a complete redesign of the memory will be required to maintain the benefits of OPCMs. In particular, SiPh links integrated with photonic components [13], [70], [71], [72], [73] should be used to convert and communicate the data from electronic units (e.g., memory controller) to the OPCM-based main memory. This area is still evolving and researchers are trying to design energy-efficient and low-latency OPCM-based architectures for emerging computing systems.

The work in [5] presented a novel OPCM-based memory architecture called "COSMOS." In this work, to make the electronic part of the memory (i.e., the memory control unit) compatible with an array of OPCM cells to replace the DRAM, an electro-opto-electrical (E-O-E) unit is required. This unit has an essential role in converting the electrical signals from the memory control unit-such as row and column addresses, Read, Write, and Erase commands-to optical signals. The converted optical signals will be then transmitted through silicon photonic links to access the OPCM cells in the architecture. Fig. 7 shows the block diagram of the memory architecture designed in [5]. Each OPCM cell in the architecture can store 5-bit data using the design in [12]. The OPCM data storage cell in this design has a structure similar to the one proposed in [74], where silicon photonic MRRs are used to access the cells and different wavelengths address the cells in different row and column addresses. In [5], a detailed analysis was performed on the impact of different parameters related to OPCM cells, including the number of bits per cell (MLC capacity), number of silicon photonic links, and OPCM capacity on the overall performance of the memory.

The work in [5] showed that the execution time for a memory with 4-bit OPCM cells and 64 silicon photonic links is much lower compared to a memory with 2-bit EPCM cells and 64 electrical links, as well as higher read and write throughput and lower average latency. Regarding the effect of the MLC capacity, it was shown that as the MLC capacity of a single OPCM cell increases, the average memory (read and write) latency decreases by about 33%. This is because of the enhanced parallelism that an OPCM cell with a higher MLC capacity can offer. The energy-per-bit for read and write in [5] was reported to be, respectively, 243 and 44.5 pJ/bit for the EPCM, and to be 40.68 and 11.6 pJ/bit for the OPCM, showing a significant enhancement in the energy efficiency of the memory. The number of silicon photonic links and the average lifetime of the OPCM were explored in [5]. It was found that increasing the number of silicon photonic links from 64 to 256 can enhance the execution time of the memory by about 5 seconds. However, this comes at a cost of higher loss and crosstalk noise related to silicon photonic links. Increasing the MLC capacity from two to eight bits can also decrease the average lifetime from 16 years to six years. Finally, a detailed analysis was presented in [5] on the area consumption of OPCMs. The work reported a significant improvement in the area consumption of a 2 GB 3D stacked



FIGURE 7. COSMOS Architecture [5].

8-bit OPCM (67.1 mm²) compared to the equivalent DDR4 counterpart (224 mm²).

D. PHASE CHANGE MATERIALS FOR PHOTONIC IN-MEMORY COMPUTING

With the increased complexity of deep learning algorithms and AI applications, there is a critical need for improving digital multiplier units in AI hardware accelerators. Indeed, matrix-vector multiplication, which is the core of every machine learning and deep learning model, is known to be the most time and energy-consuming operation in AI applications running on AI accelerators [75], [76]. Accordingly, the performance of electronic AI accelerators degrades as the complexity and the size of the models they run continue to grow. Emerging integrated photonic computing platforms have shown a great promise to perform multiplication and accumulation (MAC) operations with significantly lower latency and power consumption [75], [77], [78]. Because of using photons instead of electrons for data movement and computation, compared to conventional AI accelerators, photonic AI accelerators offer higher bandwidth and parallelism in the computation in addition to lower latency due to performing computations at light speed.

Several optical neural networks based on silicon photonic integrated circuits have been proposed, including coherent networks using MZIs [78], [79], [80], [81] and noncoherent networks using MRR banks [77], [82], [83], [84], [85], [86], [87], hence enhancing the hardware implementation of multiplication units with higher speed and lower computation energy consumption compared to electronic accelerators, photonic AI accelerators suffer from inherent limitations, including optical loss and crosstalk noise [75], large footprint ($\approx 1000 \ \mu$ m for a deep neural network with two hidden layers), and sensitivity to thermal and process variations, which result in deterioration of the system's overall performance (e.g., drop in inferencing accuracy) as the network scales up [88], [89], [90], and [91].

PCMs can be integrated with photonic AI accelerators to address some of the limitations of photonic accelerators, such as large footprint and accumulated optical losses. Some of the state-of-the-art applications of PCMs in the implementation of photonic tensor cores will be discussed in Section IV. In addition to using PCMs for photonic computing, recent work has shown that phase-change memories can be also used as part of in-memory computing units [92], [93], [94], [95]. Photonic computing units based on PCMs are beneficial due to lower static power consumption and smaller footprint [96]. The reason for this is the elimination of the bulky phase shifters in the photonic computing units (e.g. MZIs), and replacing them with smaller PCM cells that can be driven with an optical signal and without additional applied biasing to maintain the PCM state [97]. However, in data-driven applications, most of the power in the system is consumed not in the computing units but during the data transfer between processing units and the main memory due to the separation of the memory unit and processing units in Von Neumann architectures [24], [98]. This challenge still remains and it deteriorates the system's overall performance as the size and complexity of applications increase. This is one of the main motivations for using OPCMs for both computation and data storage, and this paradigm is called photonic in-memory computing [99].

The work in [100] and [101] showed that a single multilevel GST cell, which can be used as a phase-change memory, is able to perform a simple scalar multiplication. The design principles for implementing a simple PCM-based photonic multiplier are shown in Fig. 8. Based on the proposed design, a single scalar-scalar multiplication can be performed through the transmission level of the GST cell. The transmission level of the GST cell can be changed by changing the crystalline area of the GST similar to what we have seen in multilevel photonic memory cells [12]. Manipulation of the crystalline area of the GST in this design can be done through the modulation of the input optical signal's power and duration. Depending on the crystalline area of the GST cell, its transmission level, which represents the intended scalar value, varies. The GST's transmission level can act as a scalar that is multiplied by the input optical signal and propagates to the output of the cell. Note that the input optical signal to read the data from a cell must have lower energy than the programming signal that is used to set the cell to prevent a change in the transmission level of the GST.

The work in [102] presented a design for an in-memory photonic computing unit based on SOI platform integrating PCMs (i.e., GST). The schematic of this design is depicted in Fig. 9 (b). It was shown that by using slot-ridge waveguides one may increase the dynamic range of the weights to be used for in-memory computation. The main difference of the design in [102] with the one demonstrated in [100] is the enhancement made in the interaction between the mode confined in the underlying waveguide and the PCM cell. It was shown that this design is capable of storing more multilevel weights, hence an increase in the energy efficiency of the in-memory photonic computing unit.

From Figs. 9(d) and 9(f), it can be seen that the interaction between the electric field of the confined mode in the waveguide and the GST is significantly enhanced for the slot-ridge waveguide, compared to the conventional ridge waveguide



FIGURE 8. a) Scheme of reading and programming pulse of the GST as well as its geometry. b) Write pulse amorphousize some portion of the crystalline GST, leading to higher transmission levels. c) A simple scalar multiplication. d) Readout scheme with a P_{in} pulse which is unable to induce phase transition in GST because of its lower energy and duration compared to the write pulse. e) Sample of the fabricated device to perform a simple matrix-vector multiplication [100]. Permission: https://creativecommons.org/licenses/by/4.0/.

integrated with GST proposed in [100]. In addition, the optical transmission change in the slot-ridge design is increased significantly compared to the ridge design because of the enhanced interaction of the electric field and GST (see Figs. 9(a) and 9(b)). The change in the optical transmission level is an important parameter due to its crucial role in determining the MLC capacity of the OPCM cell. A higher dynamic range of the change in the optical transmission level of a single cell allows designers to store a higher number of bits per OPCM cell. The dynamic range of the optical transmission level in slot-ridge design can be even further enhanced by increasing the width of the slot. The designated design based on slot-ridge waveguides was tested by implementing a three-layer perceptron to perform a recognition task on MNIST handwritten digit dataset. An accuracy of 90.7% was achieved which was 2.6% higher than the conventional-ridge-waveguide design.

OPCM cells based on MRRs can also be used as a promising alternative for ridge-waveguide-based OPCM cells to implement photonic in-memory computing units. The work in [103] proposed a design in which GST was integrated into MRRs to implement an in-memory computing unit for spiking neural networks. By this design, the need for off-chip DRAM access to load the weights related to the implemented trained deep neural network using photonic tensor cores was eliminated. The configurability challenge for this design was also addressed by adding a bent waveguide to each synapse. The added bent waveguide can be used as the write port on the MRR-based phase-change memory cells for each synapse. In addition, the width of the write port was much smaller than the one related to MRRs. The reasoning behind such a design is to achieve an asymmetric design to have max-



FIGURE 9. Design comparison of an in-memory photonic computing unit based on phase-change materials (GST): (a) The design presented in [100]. (b) The design presented in [102]. (c) Electric field profile for GST-ridge waveguide design when GST is in the amorphous state. (d) Electric field profile for GST-slot ridge waveguide design when GST is in the amorphous state. (e) Electric field profile for GST-ridge waveguide design when GST is in the crystalline state. (f) Electric field profile for GST-slot ridge waveguide design when GST is in the crystalline state. Permission: https://creativecommons.org/licenses/by/4.0/.

imum light confinement in MRRs and less confinement in the write port to avoid any interference pattern. Using this writing scheme on the MRR-based OPCM cells made the design reconfigurable. This is becasue of the low error in the transmission in the presence of the write bent waveguide (i.e., $\approx 0.5\%$ for a separation of 300 nm between the write bent waveguide and the ring), in addition to high coupling between the bent waveguide and the ring (i.e., $\approx 70\%$ for the same separation). This made efficient writing on the MRR-based OPCM cell possible. The design in [103] was tested in the implementation of a PCM-based in-memory photonic computing unit trained on the MNIST dataset. The inferencing accuracy of 97.84% was achieved, which was really close to the inferencing accuracy of the ideal spiking neural networks at 98.34%.

Broadcast and weight is a protocol which was introduced initially in [77] to implement neuromorphic photonic processors. The broadcast-and-weight protocol uses wavelength-division multiplexing (WDM) to carry out computation in the photonic domain by using reconfigurable, continuous-valued filters called MRR banks to weigh the optical signals. Considering the design in [100], the work in [104] proposed Starlight, featuring a GST cell to exploit in-memory computing to enhance the parallelism in computation and using MRR-bank architectures in the

noncoherent photonic neural network that can be updated by using mode-wavelength division multiplexing. Starlight design showed lower power consumption and crosstalk noise compared to conventional broadcast-and-weight architectures [77], [82]. In particular, Starlight design showed an insertion loss of 0.05-0.1 dB and crosstalk noise of -50 to -30 dB for resonant and non-resonant state. The structure of a single dot-product engine that is used in Starlight is shown in Fig. 10. We can see that each kernel weight element can be translated into an optical transmission level of the GST using a write laser pulse to achieve a specific crystallization area. Programming the GST cells based on the kernel weights in Starlight was performed offline after the training of the network, as a one-time procedure. In addition, Starlight design showed that using hybrid mode-wavelength division multiplexing eliminates the need for high number of wavelengths to perform acceleration in addition to increased parallelism in the photonic in-memory computation using GST cells.

The Starlight architecture is illustrated in Fig. 11. We can see that by using multiple modes and GST-based in-memory computing units to perform matrix-vector multiplication, the number of wavelengths-and hence the number of the laser sources-can be decreased and this leads to enhanced parallelism in computation. In addition, this design strategy makes Starlight much more energy-efficient compared to conventional broadcast-and-weight architectures to carry out AI acceleration tasks [77]. In terms of power and area overhead of Starlight architecture, it was shown that using four TE modes and four wavelengths per mode (4 \times 4 \times 4 photonic acceleration unit) leads to a chip area of less than 0.4 mm^2 and consumes 0.078 W of power, which is lower than conventional noncoherent broadcast-and-weight architectures, such as DEAP [105] with a power consumption of 1.404 W. Eventually a $4 \times 4 \times 4$ Starlight design was tested on Iris classification dataset and it achieved the inferencing accuracy of 96%, which was consistent with the accuracy that was obtained during the training process.

Although embedding photonic tensor cores with PCMbased in-memory computing cells shows a promising prospect for photonic in-memory computing paradigm, two key bottlenecks of restricted bit-width and write endurance of the PCM-based photonic computing cells limit their performance. The size of the in-memory photonic computing units grows with the increase in the size of the designated datadriven applications. This necessitates numerous weight writes on cells inside the computing unit. Therefore, reduction of the lifetime and the MAC unit's accuracy is inevitable. Aging of the cells over the long run leads to the degradation of obtainable transmission ranges, and consequently, the accuracy can drop severely. The work in [54] showed that with only 30% aged cells in a photonic tensor core, the accuracy of the multiplication decreases by 35%. To alleviate this critical issue, an aging-aware optimization methodology for PCMbased in-memory photonic tensor cores, called "Elight," was presented in [54]. The work in [54] also described a



FIGURE 10. Starlight dot product engine using MRRs and integrated GST [104]. Permission: https://creativecommons.org/licenses/by/4.0/.

write-aware training process for AI applications that reduces the number of write operations (by roughly 20 times), which is the primary cause of aging. In addition, a post-training optimization procedure is provided to further reduce the number of writing operations.

E. SIMULATION OF OPCMs

A simulation algorithm is of essence to start from the device and material level to system level to simulate the OPCM cells and optimize them according to the given design goals. In this part of our survey, we briefly discuss the simulation procedure of PCM-based photonic devices. The simulation of a PCM itself is nontrivial because of the need for multidomain simulations, including molecular, optical (electrical), and thermodynamic (heat) simulations. In addition, most of the research efforts that were presented so far are dealing with a single-cell analysis. As an example, the work in [22] indicated that for a top-illuminated sample of GST with the thickness of 80 nm and length of 20 μ m, a laser pulse with a power of 3-5 mW and duration of 50-600 ns is required to obtain a fully crystalline state. The work in [56] also showed that to obtain 18% change in the transmission of a 10-nmthick and 2- μ m-long GST, about 650 pJ energy with a laser pulse with a power of 2.3 mW is needed. This implies that the laser pulse duration should be roughly 282.60 ns for a single write cycle. In the same work, it was shown that as the thickness of the GST cell increases, due to its higher optical absorption, lower energies will be needed at the input to trigger the same transition for a single write cycle. Suppose that we use 256 of such cells in an OPCM-based architecture to replace DRAMs [5]; this implies that significant optical power will be needed to address and control these cells. The required optical power can be decreased at a cost of increased write latency.

The work in [22] proposed such a platform and a simulation algorithm written in MATLAB, which is available publicly [106]. Nevertheless, the proposed simulation platform



FIGURE 11. Starlight AI accelerator architecture. a) The architecture of the tiles in the processor. b) Architecture of each tile using multiple photonic accelerator units. c) Architecture of each photonic accelerator unit using hybrid mode- and wavelength-division multiplexing to perform acceleration in parallel. d) Architecture of each photonic acceleration unit using a single mode and multiple wavelengths to perform the matrix-vector multiplication [104]. Permission: https://creativecommons.org/licenses/by/4.0/.

has two major drawbacks: 1) it is 2D, and 2) the device is top illuminated instead of using a Si₃N₄-based ridge waveguide, similar to what is proposed in [11], [53], and [100]. The simulated results were verified using experimental results. The simulation steps performed in [22] are depicted in Fig. 12. As it can be seen from Fig. 12, the simulation starts with illuminating the GST sample with a laser pulse with specific duration and power. The input optical pulse will be partially absorbed by the sample and converted to heat, leading to temperature fluctuations in the sample. The heat transfer equation must be solved using numerical approaches, such as finiteelement modeling (FEM), to obtain the temperature distribution at each time step. Based on the temperature distribution, nucleation and growth rates can be calculated. By using Cellular Automata-Nucleation and Growth model [22], the amorphous and crystalline regions can be modelled and their fractional distribution can be calculated. Eventually, from Lorenz and Fresnel equations, the transmission and reflection of the light can be calculated.

An overview of the results presented in [22] is shown in Fig. 13. Considering Fig. 13(b), the GST sample used in the simulations and experiments is top-illuminated via a laser pulse. For a constant laser duration, as the laser power increases, the diameter of the crystalline area increases. This causes an increase in the crystalline fraction of the sample, and hence a change in the optical reflectivity and transmissivity of the GST sample. As another experiment, the work in [22] suggested that the reflectivity and crystalline fraction of the GST sample increase with increasing the laser pulse duration and keeping its power constant (5.7 mW in [22]). The aforementioned trend was expected because for a constant laser pulse and increased duration, the amount of absorbed energy over time increases, leading to triggering phase transition from amorphous to crystalline state in the GST.

The simulation of GST-based photonic memory cells can also be done with computer-aided design (CAD) simulation tools, like COMSOL Multiphysics. The simulation should be performed for each time step. The phase transition model of GST can be modelled through a highly optimized Cellular Automata–Nucleation and Growth model, as it was noted in [107], [108], [109], and [110]. Unfortunately, there



FIGURE 12. GST simulation steps demonstrated in [22]. Permission: https://creativecommons.org/licenses/by/4.0/.



FIGURE 13. An overview of results presented in [22]. a) Schematic of the experimental testing unit used to verify the results. b) Verification of the simulation results by experimental results. Mean diameters of crystallised regions at different input laser powers. The dashed green line indicates the FWHM of the laser beam. c) Simulation results using a 5.4 mW laser pulse with different duration. d) Comparison of simulation and experimental results, The colorful grains are GST in crystalline state and the blue region is the GST in amorphous state. Permission: https://creativecommons.org/licenses/by/4.0/.

is no CAD simulation tool that can perform the Cellular Automata–Nucleation and Growth simulation due to its complexity and novel nature. Thus, it should be hard-coded and integrated into other multiphysics tools, such as COMSOL Multiphysics, similar to what was proposed in [56]. The work in [56] used the RF tool of COMSOL Multiphysics integrated to its HEAT TRANSFER tool to obtain the temperature distribution at each time step for a given geometry. Then, the output temperature distribution can be used by a hard-coded Cellular Automata–Nucleation and Growth code to obtain growth and

VOLUME 11, 2023

nucleation rates precisely. This procedure can be repeated until the simulation time is finished. The simulation time to reach the convergence can vary from a couple of minutes to hours depending on the simulation parameters, such as the mesh accuracy, thickness, width, and length of the cell and duration of the laser pulse.

IV. OPEN CHALLENGES AND OPPORTUNITIES

In this section, we discuss some open device- and system-level challenges and requirements that should be considered when designing phase-change memories and PCM-based devices and architectures.

A. POWER-LATENCY TRADE-OFF

The energy that is needed for setting or resetting OPCM cells depends on the PCM's geometry and the pulse peak power and its duration. As a PCM sample becomes bulkier and thicker, the absorption enhances, thus phase change can be triggered faster. However, a PCM sample cannot be thinner than 2 nm, because for an extremely thin PCM, the PCM loses its dynamic phase-change properties [10]. The energy of an optical pulse can be approximated by multiplying the pulse duration with its peak power. As the peak power reduces, the pulse duration must increase to provide enough energy to trigger a single set or reset routine via phase transition of the PCM. Typically, using optical pulses with low peak power and low duration is desired for PCM-based memory architectures. With an increase in the pulse duration and the number of OPCM cells, the average latency of the entire system will increase, and this leads to the deterioration of the memory throughput.

Increasing the peak power will cause additional heating to the nearby devices, affecting the performance of nearby

Parameter	EPCM	ReRAM	OPCM
Cell size	$\approx 4-12 \mathrm{F}^2$	$\approx 4-12 \text{ F}^2$	$\approx 500 \ nm \times 500 \ nm$
Write energy (J/bit)	$\approx 10 \text{ pJ}$	$\approx 0.1 \text{ pJ}$	60–130 pJ
Write latency (ns)	150-1000 ns	50 ns	25 ns
Read latency (ns)	50-80 ns	10 ns	25 ns
Write endurance	1e8	1e11	1e7
Retention	>10y	>10y	6-16y (depending on MLC capacity)
MLC	YES	NO	YES
3D integration	YES	YES	YES

 TABLE 4. Different features of NVM technology [5], [12], [49], [53], [111],

 [112]. (F: feature size of the lithography).

photonic devices due to the thermo-optic effect and thermal crosstalk. Increasing the peak power also results in an increase in the total power consumption of the memory system, making the system power inefficient. Therefore, optimizing the geometry of a single cell as well as that of a memory array architecture is of great importance. In addition to the optimized geometry, using a hybrid OPCM cell can help reduce the programming energy. The work in [45] suggested that using a single layer graphene (SLG) heater in the structure of the OPCM can lead to a significant 20-fold reduction $(8.7 \pm 1.7 \text{ aJnm}^3)$ in the OPCM cell programming energy density, compared to the state-of-the-art OPCM cell designs. The phase switching of the GST in the OPCM cell presented in [45] was triggered electrically by the application of a bias voltage using two gold contacts.

Because of the nonvolatile nature of PCMs, OPCM arrays offer zero static power to maintain the transmission level of the PCM cells when being used as a photonic main memory or in-memory photonic computing unit [104]. Nevertheless, to the best of our knowledge, the power-latency trade-off in OPCM-based memory systems has not been discussed in any prior work. Yet, the outlook of this paradigm can be predicted by comparing state-of-the-art NVMs (e.g., EPCMs) with DRAMs, as similar challenges may exist when using OPCMs as the main memory or in-memory computing unit. Using OPCMs for memory systems can increase the power overhead and latency of the system. The work in [5] suggested using SiPh links for communication between the memory controller and OPCM-based memory. However, using a SiPh link for such short-distance communication is energy inefficient, and requires multiple electro-optical and opto-electrical conversions per cycle [5]. This will lead to added latency and power consumption because of the limits of laser sources' wall-plug efficiency and the overhead of electro-optical/optoelectrical converters (see Fig. 7). Table 4 shows some of the key features related to the state-of-the-art NVMs. We can see from the table that in many aspects, like write energy and latency, write endurance and read latency, the performance of state-of-the-art EPCMs is even worse than ReRAMs and DRAMS. However, they offer better data retention, higher scalability, and MLC [12], [49], [111], [112]. The aforementioned limitations related to EPCMs are the main motivation for using OPCMs in future memory and computing systems. Based on the information in Table 4, any future OPCM-based memory architecture must have enhanced latency, endurance, and energy efficiency compared to those based on EPCMs.

B. ENDURANCE

The results presented in [5] show that as we increase the MLC capacity of a single OPCM cell, the average lifetime of the single OPCM cell, and hence the whole memory, decreases significantly. The writing procedure is the most critical procedure that affects the lifetime of OPCM cells. This is the main reason that in many implementations, a train of pulses is being used to write on a single OPCM cell. There is a trade-off between the average lifetime and MLC capacity of a cell. As we increase the MLC capacity, the OPCMs can be more scalable with a higher capacity, but this comes at a cost of reduced endurance. Optimization approaches were proposed to reduce the write procedures when a GST cell is being used as an in-memory computing unit. In particular, the work in [54] showed such optimization by performing a write-aware training to reduce the number of unnecessary writes when updating the weights during the training. The same work also showed a post-training optimization approach based on block-matching-based training and column-based reordering to reduce the number of redundant writes-with a negligible effect on the model's output accuracy-to increase the endurance of PCM-based photonic tensor cores [54].

C. AGING AND TRANSMISSION AND RESISTANCE DRIFT

Utilizing the MLC property of phase-change memories requires the different states in the PCMs to be well separated over time. Variation of the phase change material's state over time leads to instability in resistance and transmission. This phenomenon is called optical- and resistance-level drift and it is originated from thermal instability and optical band-gap widening of PCMs over time. The optical band-gap widening happens in PCMs in an amorphous state because of structural relaxation [113], [114] over time. This leads to a change in the optical transmission level in OPCM cells over time because of the physical nature of the drift in PCMs in the amorphous state. The instability in the state of the material level deteriorates even more over time because of aging of the cell. There are some work that tried to alleviate the resistance drift of the PCMs originating from the instability of the states. For example, the work in [115] proposed some physical-level strategies to decrease the resistance drift and increase the thermal stability of multilevel phase-change memory cells. The work in [115] related the resistance drift of the PCMs to the band-gap energy and sheet resistance of the material [116]. It was shown that this limitation can be alleviated via engineering the band gap, sheet resistance, and dielectric constant of the deposited phase change materials.

D. LOSS AND CROSSTALK NOISE

In some existing work for implementing OPCM cells, MRRs and waveguide crossings are exploited to construct, read, and write single photonic memory cell in a crossbar architecture [5], [53], [62]. Such devices suffer from coherent and noncoherent optical crosstalk noise, leading to misreading the memory in the read procedure, as well as changing

the transmission level when using higher MLC capacity for a single cell. Insertion loss can also lower a signal's amplitude used for reading multiple OPCM cells in an array of cells. These issues become even more critical when scaling memory arrays because of the accumulation in insertion loss and crosstalk noise power, leading to an increase in the number of misreads. Design optimization approaches must be used to deal with this issue to make OPCM-based memory architectures more scalable and resilient to optical loss and crosstalk noise [117], [118], [119], [120], [121].

Despite a wide range of applications of GST in constructing OPCM cells, this material shows a high insertion loss in photonic memory applications [61], especially when it is in the crystalline state due to its high extinction coefficient (see Fig. 2(c)). This limits the scalability of the architectures based on GST. One possible solution to address this issue is to use a new class of PCMs, called GSST. GSST, which has been studied for PCM-based photonic applications, shows superior performance in terms of thermal stability-through doping Se atoms and excess Ge atoms, which leads to an increase of the T_g [42], [122], [123]—and lower material loss contrast compared to GST (because of lower $\Delta \kappa$) [34], [122], [123]. The work in [21] suggested that different figures of merits (FOMs) of FOM₁ = $\Delta n / \Delta \kappa_a$ and FOM₂ = $\Delta n / \Delta \kappa_c$ can be used to quantify the performance of OPCMs. For memory purposes, a drastic change in the refractive index (Δn) upon the phase change in the OPCM cell is required. However, a large extinction coefficient change (κ) in the amorphous and crystalline state imposes excessive insertion loss to the OPCM cell due to material absorption (see II-D). The work in [21] showed that for GST, $FOM_1 = 109.72$, which is lower than that for GSST with $FOM_1 = 116.73$ at 1550 nm. This comparison can be interpreted as a higher refractive index and lower material loss that GSST in amorphous state offers compared to GST in the same state at 1550 nm. The same comparison can be made for crystalline state using FOM₂. We can see from [21] that for GSST, $FOM_2 = 4.17$, which again is higher than GST with $FOM_2 = 2.52$ at 1550 nm. Therefore, GSST applications to construct OPCMs can be expanded to obtain photonic memory arrays and computing units to alleviate the thermal instability and high material loss penalty in GST.

E. THERMAL SENSITIVITY

Thermal sensitivity is another major limitation to be considered, which can lead to transmission drift in a single cell. In [36], an analysis was carried out to model the thermo-optic coefficient in a PCM. An increase in the optical power dissipation in nearby optical components can be experienced by OPCM cells in the form of thermal variations (i.e., thermal crosstalk). Thermal variations may not trigger the phase transition but they can change the refractive index of the PCM in a particular state, leading to an increase in the number of misreads in OPCMs. The work in [36] presented analytical models for thermo-optic coefficient of GST. The presented models can be used to further optimize the performance of GST-based OPCM cells [34], [36]. In addition, the work in [124] presented an experimental study on the impact of temperature variation and transmission drift in OPCMs employed for in-memory computing units. The presented model in [124] can be used to optimize the performance of deep neural networks when they are being implemented using OPCM in-memory computing units.

F. FLICKER NOISE AND UNCERTAINTY OF THE INITIAL STATE

We learned that EPCMs can be good candidates to carry out analog in-memory computation for deep learning and AI applications. However, they suffer from flicker noise as random electron traps can be identified in the PCM's lattice. Flicker noise for EPCMs is even higher when the PCM used in the phase-change memory structure is in the amorphous state [125]. In addition to flicker noise, the PCMs used in the same phase-change memory cells respond differently to the same programming pulse. This issue will lead to inaccuracy and dispersion of OPCMs and EPCMs when being used as an in-memory computing unit. The work in [125] demonstrated that the aforementioned issues can be alleviated by using an optimized iterative programming approach. They showed that by using this programming technique, the conductance spread is under 14% and the relative drift is under 15%, with the relative noise being less than 9% for 90% of the cells.

G. COMPLEXITY AND FOOTPRINT OVERHEAD

We observed in [5] that we cannot simply replace the EPCMs with OPCMs in a memory architecture. Simply replacing the EPCMs with OPCMs will lead to thermal, read and write latency, and energy issues because of the incompatibility of OPCMs with integrated memory architectures. Consequently, additional photonic I/O and circuitry, such as E-O-E units, will be needed to convert the memory controller commands into optical signals. Moreover, a large number of silicon photonic links is required to replace the electrical links to convey the information from the memory controller to the OPCMs and vice versa. This requires using a vast number of active devices, such as electro-optical modulators, high-speed integrated photodetectors [126], and additional control units to optically address the OPCM cells. This makes the memory architecture more susceptible to optical loss and crosstalk noise, increases the power consumption and footprint, and diminishes the memory bandwidth [127]. In addition, the design of the E-O-E unit should be optimized to minimize the access time to the memory to read and write the cells. Furthermore, to deal with the device- and material-level limitations, the OPCM cells should be optimized to have minimal power consumption and latency.

H. FABRICATION-PROCESS VARIATIONS

Fabrication-process variations is a known issue in silicon photonic integrated circuits, which necessitate additional efforts to design devices that are resilient to variations [97], [128], [129], [130], [131]. In the state-of-the-art designs for OPCM cells, the thickness of the PCM was considered extremely thin (i.e., ≈ 10 nm), and hence a slight variation in the thickness of the PCM sample leads to changes in the power absorption, transmission levels, and stored data. This limitation can increase the number of misreads and the bit-error rate (BER) in OPCM-based memory systems [5]. The effect on the latency and power consumption can be minor, but it will be accumulated as we scale the memory architecture. Therefore, OPCM cells should be designed and optimized to be resilient to fabrication imperfections. Mathematical models can be used to model the attenuation coefficient against the sample geometrical parameters and variations, like thickness and length of the OPCM cells, in which the GST is fabricated on top of a Si_3N_4 ridge waveguide [132]. The attenuation coefficient can be used as a figure of merit to design an OPCM cell for data storage and photonic computing which is resilient to fabrication-process variations.

I. SCALABILITY

Scalability is of importance when it comes to photonic data storage technologies. As the dimension of a single OPCM cell decreases, its properties also change according to its size reduction. Research shows that as the phase change material's film thickness decreases, T_g , optical band gap, incubation time, resistivity, refractive index, and extinction coefficient for most wavelengths increases [10], [133], [134], [135], [136], [137]. It was also shown that the crystallization speed would increase as the thickness of the PCM decreases. Overall, the thickness of a PCM in OPCMs cannot be smaller than 2 nm becasue the material will lose its dynamic properties stemming from phase transition.

J. SECURITY

Security is another challenge that stems from the aging issue in phase-change memories. Nonvolatile phase-change memory cells offer a limited capacity of write procedures. With an increase in the number of writes over time, the memory cell ages, and hence it is prone to write attacks [138]. The data which has been written on phase-change memory cells will remain even after power-off. The reason for this is because of using the physical state of the material to store the data. This will increase the threat of data accessibility by a malicious agent. In addition to the aforementioned threats, a bus snooping attack is another possible security threat in phase-change memories where a third agent can access the data which is being communicated over a chip. Different mapping and encryption approaches can be utilized to overcome this challenge in memory architectures based on phase-change memories.

K. PHASE CHANGE MATERIALS FOR PHOTONIC NETWORKS

In addition to the application of PCMs for implementing photonic memory architectures and photonic in-memory computing units, PCMs can also be integrated into conventional photonic devices to realize programmable photonic devices for photonic networks. For example, PCMs like Sb_2Se_3 can be integrated into passive SiPh directional couplers (DCs) to realize a tunable photonic switching cell [38], [60]. By controlling the phase state of Sb_2Se_3 , the portion of an optical signal that can be coupled from one waveguide to another in the DC can be controlled. Lower loss and crosstalk can be offered by integrating Sb_2Se_3 into SiPh DCs compared to GST and Sb_2Se_3 [19]. This was the main motivation for using Sb_2Se_3 in [38].

MZIs are critical photonic devices that can be used in a variety of applications, such as silicon photonic switching networks [150], [151], [152] and electro-optic modulators [153] for optical communication platforms. A single MZI can be constructed by connecting two DCs and two straight waveguides. Therefore, to realize MZIs with low loss and crosstalk noise, DCs should have optimal design and performance. The work in [38] demonstrated that tunable DCs with a low insertion loss (<1.5 dB) and crosstalk noise (-20 to -40 dB) can be designed to construct MZIs using Sb₂Se₃. In addition, it was shown that such a design has a fast switching time (50 ns for GST and 78 ns for Sb₂Se₃) when using the DC as a photonic switching element.

The work in [141] and [142] proposed a design utilizing PCMs (Sb₂Se₃) to implement phase shifters to realize specific optical phase delays. PCMs can be also used to tune the performance of MRRs in optical switching networks based on MRRs [154]. The undesired resonance shift in MRRs (e.g., due to thermal or process variations [128]) is known as one of the main constraints in using MRRs for wavelength-selective photonic switching elements. The work in [154] showed that GST can be used to tune the resonant wavelength in race-track MRRs to eliminate the need for power-hungry frequency response adjustments, using, for example, thermal tuning. This makes the photonic switching elements much more power efficient and more resilient to thermal and fabricationprocess variations. PCM-based switching elements can also be used to implement 2.5D SiPh interposer networks. For example, the work in [155] showed that 2.5D interposerbased photonic networks can utilize phase change materials to achieve reconfigurable bandwidth and power efficient communication in 2.5D chiplet systems. Although the application of PCMs in photonic devices is promising, more research is required to further enhance the performance of PCMs considering photonic network requirements.

L. PHASE CHANGE MATERIALS FOR PHOTONIC COMPUTING

PCMs can be also used to construct photonic computing units to accelerate the performance of AI accelerator platforms. An array of MZIs with integrated phase shifters can be used to perform matrix-vector multiplication at light speed using a single wavelength to implement coherent photonic neural networks for high-speed optical AI accelerators [139], [156]. The DCs designed in [38] can be also used to construct MZIs with low insertion loss and crosstalk noise

Reference	Application	Material	Stimulation
[38], [60]	Tunable directional couplers	Sb ₂ S ₃	Optical
[140]	Tunable MZIs	Sb ₂ S ₃	Optical/Electrical
[141]	Tunable race-track MRR	GST	Optical
[142], [143]	Photonic phase shifters	Sb ₂ Se ₃	Electrical
[144]	Application of silicon and silicon nitride in PCM-based PICs	—	-
[53], [62], [63]	Photonic data storage	GST	Optical
[6], [47], [48]	Data storage	GST	Electrical
[61]	Photonic data storage	GSST	Optical
[36]	Thermal-aware design optimization of photonic data storage cells	GST	Optical
[64]	Photonic switches	GSST	Optical
[57]	Photonic logic gates	GST	Optical
[65]	Photonic switches and data storage cells	GST	Optical
[11], [12]	Multilevel photonic data storage cell	GST	Optical
[67]	Photonic switches	GST	Electrical
[5]	OPCM-based memory architecture	GST	Optical
[67], [68], [145]	Plasmonically enhanced photonic data storage cells	GST	Optical
[146]	Electrically driven photonic computing units	GST	Electrical
[92], [93], [100], [101], [147]–[149]	Optically driven photonic computing units	GST	Optical
[54]	Performance optimization of photonic in-memory computing units	GST	Optical
[92]–[95], [99]–[104], [150]	Photonic in-memory computing units	GST	Optical/Electrical
[45]	Reducing the programming energy using a graphene layer	GST	Optical/Electrical

TABLE 5.	Summery of	f different	applications	of PCMs in	integrated	photonics.	(PIC:	photonic	integrated	circuits)
----------	------------	-------------	--------------	------------	------------	------------	-------	----------	------------	-----------

for coherent photonic neural networks [75]. In addition, considering the design in Fig. 8, the proposed PCM-based cell can be extended to perform matrix-vector multiplication in the optical domain. Compared to conventional photonic accelerators (e.g., those based on MZI arrays [75], [156] or MRR banks [77]), the benefits of this design are lower loss and accumulated crosstalk noise, elimination of bulky metallic contacts required for the phase shifters, and compact footprint.

Considering a single GST cell for photonic in-memory computing, photonic tensor cores can be designed based on wavelength-division multiplexing by using an array of MRRs integrated with phase-change memories to perform computationally expensive matrix-vector multiplication with zero static power and high speed [145], to outperform graphic processing units (GPUs). As another example, an architecture based on PCMs for photonic tensor cores was proposed in [146] and [147]. The difference of this design compared to [145] is that the PCM used in cells can be controlled via optical laser pulses. The design presented in [146] and [147] was able to operate at the speed of two tera-MAC operations per second, i.e., two trillion (10^{12}) MAC operations per second. Despite their novel approach to implement an integrated PCM-based photonic tensor core processing unit, the proposed design lacks a second unit for reference computation. The work in [148] addressed this problem by integrating a second GST cell on each photonic computing element to perform reference computation at the same time. Research in this area is still in progress to enhance the performance of the PCM-based photonic tensor cores to alleviate the scalability limitations originating from aging of the cells [157] and coherent crosstalk and insertion loss, and to store kernel weights and efficiently reconfigure noncoherent AI accelerators [84], [85], [86], [87].

V. CONCLUSION

Different from existing survey papers on photonic memories, this paper surveyed the fundamentals of PCMs from the material level to the system level for contemporary PCM-based photonic memories. A complete overview of the optical, thermal, and electrical properties of PCMs was presented. In addition, we reviewed the application of PCMs in the implementation of photonic memories and the benefits they bring to emerging many-core computing systems. Some applications of PCMs in-memory architectures and for in-memory computing units were presented as well as optimization approaches to alleviate the limitations related to the endurance caused by a high number of writes on photonic processing units, like PCM-based photonic tensor cores. Furthermore, we showed a simulation procedure for PCM-based photonic devices using commercial tools. Open challenges and opportunities in the design and fabrication of phase-change-material-based photonic devices were discussed, which can be used to further advance emerging photonic computing and communication system architectures. Table 5 summarizes the different applications discussed in this survey, from storage to non-storage photonic devices.

REFERENCES

- C. Lefurgy, K. Rajamani, F. Rawson, W. Felter, M. Kistler, and T. W. Keller, "Energy management for commercial servers," *IEEE Comput.*, vol. 36, no. 12, pp. 39–48, Dec. 2003.
- [2] S. K. Kim and M. Popovici, "Future of dynamic random-access memory as main memory," *MRS Bull.*, vol. 43, no. 5, pp. 334–339, 2018.
- [3] O. Mutlu, "Memory scaling: A systems architecture perspective," in Proc. 5th IEEE Int. Memory Workshop, May 2013, pp. 21–25.
- [4] J. Boukhobza, S. Rubini, R. Chen, and Z. Shao, "Emerging NVM: A survey on architectural integration and research challenges," ACM Trans. Des. Autom. Electron. Syst., vol. 23, no. 2, pp. 1–32, Mar. 2018.
- [5] A. Narayan, Y. Thonnart, P. Vivet, A. Coskun, and A. Joshi, "Architecting optically controlled phase change memory," *ACM Trans. Archit. Code Optim.*, vol. 19, no. 4, pp. 1–26, Dec. 2022.

- [6] F. Bedeschi, R. Fackenthal, C. Resta, E. M. Donze, M. Jagasivamani, E. Buda, F. Pellizzer, D. Chow, A. Cabrini, G. M. A. Calvi, R. Faravelli, A. Fantini, G. Torelli, D. Mills, R. Gastaldi, and G. Casagrande, "A multilevel-cell bipolar-selected phase-change memory," in *Proc. IEEE Int. Solid-State Circuits Conf.*, Feb. 2008, pp. 428–625.
- [7] Y. Chen, "ReRAM: History, status, and future," *IEEE Trans. Electron Devices*, vol. 67, no. 4, pp. 1420–1433, Apr. 2020.
- [8] G. W. Burr, M. J. Breitwisch, M. Franceschini, D. Garetto, K. Gopalakrishnan, B. Jackson, B. Kurdi, C. Lam, L. A. Lastras, A. Padilla, and B. Rajendran, "Phase change memory technology," *J. Vac. Sci. Technol. B, Microelectron. Process. Phenom.*, vol. 28, no. 2, pp. 223–262, 2010.
- [9] T. Nirschl et al., "Write strategies for 2 and 4-bit multi-level phase-change memory," in *IEDM Tech. Dig.*, Dec. 2007, pp. 461–464.
- [10] S. Raoux, "Phase change materials," Annu. Rev. Mater. Res., vol. 39, pp. 25–48, Aug. 2009.
- [11] C. Ríos, M. Stegmaier, C. D. Wright, W. H. P. Pernice, and H. Bhaskaran, "Multi-level storage in non-volatile phase-change nanophotonic memories," in *Proc. IEEE Photon. Conf. (IPC)*, Oct. 2016, pp. 408–409.
- [12] X. Li, N. Youngblood, C. Ríos, Z. Cheng, C. D. Wright, W. H. Pernice, and H. Bhaskaran, "Fast and reliable storage using a 5 bit, nonvolatile photonic memory cell," *Optica*, vol. 6, no. 1, pp. 1–6, 2019.
- [13] S. Pasricha and M. Nikdast, "A survey of silicon photonics for energyefficient manycore computing," *IEEE Des. Test*, vol. 37, no. 4, pp. 60–81, Aug. 2020.
- [14] S. Pasricha and N. Dutt, On-Chip Communication Architectures: System on Chip Interconnect. San Mateo, CA, USA: Morgan Kaufmann, 2010.
- [15] A. Shafiee, S. Pasricha, and M. Nikdast, "Silicon photonics for future computing systems," in *Wiley Encyclopedia for Electrical Engineering*, Dec. 2022, pp. 1–26. [Online]. Available: https://onlinelibrary. wiley.com/action/showCitFormats?doi=10.1002%2F047134608X.W8435
- [16] M. Nikdast, S. Pasricha, G. Nicolescu, A. Seyedi, and D. Liang, *Silicon Photonics for High-Performance Computing and Beyond*. Boca Raton, FL, USA: CRC Press, 2021.
- [17] N. Margalit, C. Xiang, S. M. Bowers, A. Bjorlin, R. Blum, and J. E. Bowers, "Perspective on the future of silicon photonics and electronics," *Appl. Phys. Lett.*, vol. 118, no. 22, May 2021, Art. no. 220501.
- [18] Z. Fang, R. Chen, J. Zheng, and A. Majumdar, "Non-volatile reconfigurable silicon photonics based on phase-change materials," *IEEE J. Sel. Topics Quantum Electron.*, vol. 28, no. 3, pp. 1–17, Jun. 2022.
- [19] M. Wuttig, H. Bhaskaran, and T. Taubner, "Phase-change materials for non-volatile photonic applications," *Nature Photon.*, vol. 11, pp. 465–476, Aug. 2017.
- [20] M. S. Nisar, X. Yang, L. Lu, J. Chen, and L. Zhou, "On-chip integrated photonic devices based on phase change materials," *Photonics*, vol. 8, no. 6, 2021, p. 205.
- [21] R. Chen, Z. Fang, F. Miller, H. Rarick, J. E. Fröch, and A. Majumdar, "Opportunities and challenges for large-scale phase-change material integrated electro-photonics," ACS Photon., vol. 9, no. 10, pp. 3181–3195, Oct. 2022.
- [22] Y. Wang, J. Ning, L. Lu, M. Bosman, and R. E. Simpson, "A scheme for simulating multi-level phase change photonics materials," *npj Comput. Mater.*, vol. 7, no. 1, pp. 1–10, Nov. 2021.
- [23] A. L. Lacaita and D. J. Wouters, "Phase-change memories," *Phys. Status Solidi*, vol. 205, no. 10, pp. 2281–2297, Oct. 2008.
- [24] N. P. Jouppi et al., "In-datacenter performance analysis of a tensor processing unit," in *Proc. ACM/IEEE 44th Annu. Int. Symp. Comput. Archit.* (ISCA), Jun. 2017, pp. 1–12.
- [25] H. Zhang, L. Zhou, L. Lu, J. Xu, N. Wang, H. Hu, B. A. Rahman, Z. Zhou, and J. Chen, "Miniature multilevel optical memristive switch using phase change material," *ACS Photon.*, vol. 6, no. 9, pp. 2205–2212, Aug. 2019.
- [26] M. Wuttig and N. Yamada, "Phase-change materials for rewriteable data storage," *Nature Mater.*, vol. 6, pp. 824–832, Nov. 2007.
- [27] J. Wang, L. Wang, and J. Liu, "Overview of phase-change materials based photonic devices," *IEEE Access*, vol. 8, pp. 121211–121245, 2020.
- [28] M. Le Gallo and A. Sebastian, "An overview of phase-change memory device physics," J. Phys. D, Appl. Phys., vol. 53, no. 21, May 2020, Art. no. 213002.
- [29] M. Wuttig and S. Raoux, "The science and technology of phase change materials," *Zeitschrift für Anorganische Allgemeine Chemie*, vol. 638, no. 15, pp. 2455–2465, 2012.
- [30] Y. Zhang, C. Ríos, M. Y. Shalaginov, M. Li, A. Majumdar, T. Gu, and J. Hu, "Myths and truths about optical phase change materials: A perspective," *Appl. Phys. Lett.*, vol. 118, no. 21, May 2021, Art. no. 210501.

- [31] M. Kuwahara, O. Suzuki, Y. Yamakawa, N. Taketoshi, T. Yagi, P. Fons, T. Fukaya, J. Tominaga, and T. Baba, "Temperature dependence of the thermal properties of optical memory materials," *Jpn. J. Appl. Phys.*, vol. 46, no. 6B, pp. 3909–3911, Jun. 2007.
- [32] W. P. Risk, C. T. Rettner, and S. Raoux, "Thermal conductivities and phase transition temperatures of various phase-change materials measured by the 3ω method," *Appl. Phys. Lett.*, vol. 94, no. 10, Mar. 2009, Art. no. 101906.
- [33] J. Gosciniak, "Waveguide-integrated plasmonic photodetectors and activation function units with phase change materials," 2021, arXiv:2109.13562.
- [34] K. Aryana, Y. Zhang, J. A. Tomko, M. S. B. Hoque, E. R. Hoglund, D. H. Olson, J. Nag, J. C. Read, C. Ríos, J. Hu, and P. E. Hopkins, "Suppressed electronic contribution in thermal conductivity of Ge₂Sb₂Se₄Te," *Nature Commun.*, vol. 12, no. 1, pp. 1–9, Dec. 2021.
- [35] (2001). Silicon Nitride (Si₃N₄) Properties and Applications. [Online]. Available: https://www.azom.com/properties.aspx?ArticleID=53
- [36] M. Stegmaier, C. Ríos, H. Bhaskaran, and W. H. P. Pernice, "Thermooptical effect in phase-change nanophotonics," ACS Photon., vol. 3, no. 5, pp. 828–835, 2016.
- [37] K. Kohary and C. D. Wright, "Electric field induced crystallization in phase-change materials for memory applications," *Appl. Phys. Lett.*, vol. 98, no. 22, May 2011, Art. no. 223102.
- [38] T. Y. Teo, M. Krbal, J. Mistrik, J. Prikryl, L. Lu, and R. E. Simpson, "Comparison and analysis of phase change materials-based reconfigurable silicon photonic directional couplers," *Opt. Mater. Exp.*, vol. 12, no. 2, pp. 606–621, 2022.
- [39] H. J. Kim, J.-W. Sohn, N. Hong, C. Williams, and W. Humphreys, "PCM-Net: A refractive index database of chalcogenide phase change materials for tunable nanophotonic device modelling," *J. Phys., Photon.*, vol. 3, no. 2, Apr. 2021, Art. no. 024008.
- [40] G. Ghione, Semiconductor Devices for High-Speed Optoelectronics. Cambridge, U.K.: Cambridge Univ. Press, 2009, vol. 116.
- [41] Y. Qu, Q. Li, K. Du, L. Cai, J. Lu, and M. Qiu, "Dynamic thermal emission control based on ultrathin plasmonic metamaterials including phasechanging material GST," *Laser Photon. Rev.*, vol. 11, no. 5, Sep. 2017, Art. no. 1700091.
- [42] Y. Zhang, C. Fowler, J. Liang, B. Azhar, M. Y. Shalaginov, S. Deckoff-Jones, S. An, J. B. Chou, C. M. Roberts, V. Liberman, M. Kang, C. Ríos, K. A. Richardson, C. Rivero-Baleine, T. Gu, H. Zhang, and J. Hu, "Electrically reconfigurable non-volatile metasurface using low-loss optical phase-change material," *Nature Nanotechnol.*, vol. 16, no. 6, pp. 661–666, Jun. 2021.
- [43] C. Ríos, Y. Zhang, Q. Du, C. C. Popescu, M. Shalaginov, P. Miller, C. Roberts, M. Kang, K. Richardson, S. An, and C. Fowler, "Electricallyswitchable foundry-processed phase change photonic devices," in *Proc. SPIE*, vol. 11796, 2021, Art. no. 117961Z.
- [44] C. Ríos, Q. Du, Y. Zhang, M. Shalaginov, P. Miller, C. Roberts, M. Kang, K. A. Richardson, T. Gu, S. Vitale, and J. Hu, "On-chip electrothermal switching of low-loss phase change materials for nonvolatile programmable photonic circuits," in *Proc. Conf. Lasers Electro-Opt. Eur. Eur. Quantum Electron. Conf. (CLEO/Europe-EQEC)*, Jun. 2021, pp. 1–2.
- [45] Z. Fang, R. Chen, J. Zheng, A. I. Khan, K. M. Neilson, S. J. Geiger, D. M. Callahan, M. G. Moebius, A. Saxena, M. E. Chen, C. Rios, J. Hu, E. Pop, and A. Majumdar, "Ultra-low-energy programmable non-volatile silicon photonics based on phase-change materials with graphene heaters," *Nature Nanotechnol.*, vol. 17, no. 8, pp. 842–848, Aug. 2022.
- [46] B. C. Lee, E. Ipek, O. Mutlu, and D. Burger, "Architecting phase change memory as a scalable dram alternative," in *Proc. 36th Annu. Int. Symp. Comput. Archit.*, Jun. 2009, pp. 2–13.
- [47] I. G. Thakkar and S. Pasricha, "DyPhase: A dynamic phase change memory architecture with symmetric write latency and restorable endurance," *IEEE Trans. Comput.-Aided Design Integr. Circuits Syst.*, vol. 37, no. 9, pp. 1760–1773, Sep. 2018.
- [48] A. Cabrini, S. Braga, A. Manetto, and G. Torelli, "Voltage-driven multilevel programming in phase change memories," in *Proc. IEEE Int. Work-shop Memory Technol., Design, Test.*, Sep. 2009, pp. 3–6.
- [49] H.-K. Liu, D. Chen, H. Jin, X.-F. Liao, B. He, K. Hu, and Y. Zhang, "A survey of non-volatile main memory technologies: State-of-the-arts, practices, and future directions," *J. Comput. Sci. Technol.*, vol. 36, no. 1, pp. 4–32, Jan. 2021.
- [50] Q. Wang, G. Niu, W. Ren, R. Wang, X. Chen, X. Li, Z. Ye, Y. Xie, S. Song, and Z. Song, "Phase change random access memory for neuroinspired computing," *Adv. Electron. Mater.*, vol. 7, no. 6, Jun. 2021, Art. no. 2001241.

- [51] L. Jiang, B. Zhao, J. Yang, and Y. Zhang, "A low power and reliable charge pump design for phase change memories," ACM SIGARCH Comput. Archit. News, vol. 42, no. 3, pp. 397–408, Oct. 2014.
- [52] C. Jason Xue, G. Sun, Y. Zhang, J. Joshua Yang, Y. Chen, and H. Li, "Emerging non-volatile memories: Opportunities and challenges," in *Proc. 9th IEEE/ACM/IFIP Int. Conf. Hardw./Softw. Codesign Syst. Synth.* (CODES+ISSS), Oct. 2011, pp. 325–334.
- [53] C. Ríos, M. Stegmaier, P. Hosseini, D. Wang, T. Scherer, C. D. Wright, H. Bhaskaran, and W. H. Pernice, "Integrated all-photonic non-volatile multi-level memory," *Nature Photon.*, vol. 9, pp. 725–732, Sep. 2015.
- [54] H. Zhu, J. Gu, C. Feng, M. Liu, Z. Jiang, R. T. Chen, and D. Z. Pan, "ELight: Enabling efficient photonic in-memory neurocomputing with life enhancement," in *Proc. 27th Asia South Pacific Design Autom. Conf. (ASP-DAC)*, Jan. 2022, pp. 332–338.
- [55] J. Parra, I. Olivares, A. Brimont, and P. Sanchis, "Toward nonvolatile switching in silicon photonic devices," *Laser Photon. Rev.*, vol. 15, no. 6, Jun. 2021, Art. no. 2000501.
- [56] N. Youngblood, C. Rios, E. Gemo, J. Feldmann, Z. Cheng, A. Baldycheva, W. H. Pernice, C. D. Wright, and H. Bhaskaran, "Tunable volatility of Ge₂Sb₂Te₅ in integrated photonics," *Adv. Funct. Mater.*, vol. 29, no. 11, 2019, Art. no. 1807571.
- [57] Z. Cheng, C. Ríos, N. Youngblood, C. D. Wright, W. H. P. Pernice, and H. Bhaskaran, "Device-level photonic memories and logic applications using phase-change materials," *Adv. Mater.*, vol. 30, no. 32, Aug. 2018, Art. no. 1802435.
- [58] T. Alexoudi, G. T. Kanellos, and N. Pleros, "Optical RAM and integrated optical memories: A survey," *Light, Sci. Appl.*, vol. 9, no. 1, pp. 1–16, May 2020.
- [59] X. Li, N. Youngblood, Z. Cheng, S. G.-C. Carrillo, E. Gemo, W. H. Pernice, C. D. Wright, and H. Bhaskaran, "Experimental investigation of silicon and silicon nitride platforms for phase-change photonic in-memory computing," *Optica*, vol. 7, no. 3, pp. 218–225, 2020.
- [60] Z. Fang, J. Zheng, P. Xu, S. Deshmukh, E. Pop, and A. Majumdar, "Phase change material integrated silicon photonics: GST and beyond," in *Proc. SPIE*, vol. 11276, 2020, Art. no. 1127602.
- [61] Y. Zhang et al., "Broadband transparent optical phase change materials for high-performance nonvolatile photonics," *Nature Commun.*, vol. 10, no. 1, pp. 1–9, Sep. 2019.
- [62] C. Rios, P. Hosseini, C. D. Wright, H. Bhaskaran, and W. H. P. Pernice, "On-chip photonic memory elements employing phase-change materials," *Adv. Mater.*, vol. 26, no. 9, pp. 1372–1377, Mar. 2014.
- [63] J. Feldmann, N. Youngblood, X. Li, C. D. Wright, H. Bhaskaran, and W. H. P. Pernice, "Integrated 256 cell photonic phase-change memory with 512-bit capacity," *IEEE J. Sel. Topics Quantum Electron.*, vol. 26, no. 2, pp. 1–7, Mar. 2020.
- [64] Y. Zhang, Q. Zhang, J. B. Chou, J. Li, C. Roberts, M. Kang, C. Gonçalves, R. Soref, C. Ríos, M. Shalaginov, K. Richardson, T.Gu, V. Liberman, and J. Hu, "Designing nonvolatile integrated photonics with low-loss optical phase change materials," in *Proc. SPIE*, vol. 11081, 2019, Art. no. 110811S.
- [65] C. Rios, M. Stegmaier, Z. Cheng, N. Youngblood, C. D. Wright, W. H. Pernice, and H. Bhaskaran, "Controlled switching of phase-change materials by evanescent-field coupling in integrated photonics," *Opt. Mater. Exp.*, vol. 8, no. 9, pp. 2455–2470, Sep. 2018.
- [66] M. A. Mahdian, M. Nikoufard, and F. Soleimannezhad, "Effect of etching depth on the performance of InP-based hybrid plasmonic waveguides," *AEU Int. J. Electron. Commun.*, vol. 126, Nov. 2020, Art. no. 153403.
- [67] E. Gemo, S. G.-C. Carrillo, C. R. De Galarreta, A. Baldycheva, H. Hayat, N. Youngblood, H. Bhaskaran, W. H. Pernice, and C. D. Wright, "Plasmonically-enhanced all-optical integrated phase-change memory," *Opt. Exp.*, vol. 27, no. 17, pp. 24724–24737, 2019.
- [68] W. Sun, Y. Lu, L. Miao, and Y. Zhang, "All-optical phase-change memory with improved performance by plasmonic effect," *Photonics*, vol. 9, no. 3, p. 132, Feb. 2022.
- [69] Z. Li, R. Zhou, and T. Li, "Exploring high-performance and energy proportional interface for phase change memory systems," in *Proc. IEEE 19th Int. Symp. High Perform. Comput. Archit. (HPCA)*, Feb. 2013, pp. 210–221.
- [70] M. Nikdast, G. Nicolescu, J. Trajkovic, and O. Liboiron-Ladouceur, "Chip-scale silicon photonic interconnects: A formal study on fabrication non-uniformity," *J. Lightw. Technol.*, vol. 34, no. 16, pp. 3682–3695, Aug. 15, 2016.
- [71] F. G. de Magalhaes, R. Priti, M. Nikdast, F. Hessel, O. Liboiron-Ladouceur, and G. Nicolescu, "Design and modelling of a low-latency centralized controller for optical integrated networks," *IEEE Commun. Lett.*, vol. 20, no. 3, pp. 462–465, Mar. 2016.

- [72] X. Wu, J. Xu, Y. Ye, X. Wang, M. Nikdast, Z. Wang, and Z. Wang, "An inter/intra-chip optical network for manycore processors," *IEEE Trans. Very Large Scale Integr. (VLSI) Syst.*, vol. 23, no. 4, pp. 678–691, Apr. 2014.
- [73] M. Nikdast, J. Xu, L. H. Duong, X. Wu, Z. Wang, X. Wang, and Z. Wang, "Fat-tree-based optical interconnection networks under crosstalk noise constraint," *IEEE Trans. Very Large Scale Integr. (VLSI) Syst.*, vol. 23, no. 1, pp. 156–169, Jan. 2015.
- [74] J. Feldmann, M. Stegmaier, N. Gruhler, C. Ríos, H. Bhaskaran, C. D. Wright, and W. H. P. Pernice, "Calculating with light using a chip-scale all-optical abacus," *Nature Commun.*, vol. 8, no. 1, pp. 1–8, Nov. 2017.
- [75] A. Shafiee, S. Banerjee, K. Chakrabarty, S. Pasricha, and M. Nikdast, "LoCI: An analysis of the impact of optical loss and crosstalk noise in integrated silicon-photonic neural networks," in *Proc. Great Lakes Symp. VLSI*, Jun. 2022, pp. 351–355.
- [76] A. N. Tait, "Quantifying power in silicon photonic neural networks," *Phys. Rev. A, Gen. Phys.*, vol. 17, no. 5, May 2022, Art. no. 054029.
- [77] A. N. Tait, M. A. Nahmias, B. J. Shastri, and P. R. Prucnal, "Broadcast and weight: An integrated network for scalable photonic spike processing," *J. Lightw. Technol.*, vol. 32, no. 21, pp. 3427–3439, Nov. 1, 2014.
- [78] F. Shokraneh, S. Geoffroy-Gagnon, M. S. Nezami, and O. Liboiron-Ladouceur, "A single layer neural network implemented by a 4 × 4 MZI-based optical processor," *IEEE Photon. J.*, vol. 11, no. 6, pp. 1–12, Dec. 2019.
- [79] M. Reck, A. Zeilinger, H. J. Bernstein, and P. Bertani, "Experimental realization of any discrete unitary operator," *Phys. Rev. Lett.*, vol. 73, no. 1, pp. 58–61, Jul. 1994.
- [80] W. R. Clements, P. C. Humphreys, B. J. Metcalf, W. S. Kolthammer, and I. A. Walmsley, "Optimal design for universal multiport interferometers," *Optica*, vol. 3, no. 12, pp. 1460–1465, 2016.
- [81] F. Shokraneh, M. S. Nezami, and O. Liboiron-Ladouceur, "Theoretical and experimental analysis of a 4 × 4 reconfigurable MZI-based linear optical processor," *J. Lightw. Technol.*, vol. 38, no. 6, pp. 1258–1267, Mar. 15, 2020.
- [82] A. N. Tait, T. F. de Lima, E. Zhou, A. X. Wu, M. A. Nahmias, B. J. Shastri, and P. R. Prucnal, "Neuromorphic photonic networks using silicon photonic weight banks," *Sci. Rep.*, vol. 7, no. 1, pp. 1–10, Aug. 2017.
- [83] B. J. Shastri, A. N. Tait, T. Ferreira de Lima, W. H. Pernice, H. Bhaskaran, C. D. Wright, and P. R. Prucnal, "Photonics for artificial intelligence and neuromorphic computing," *Nature Photon.*, vol. 15, no. 2, pp. 102–114, Feb. 2021.
- [84] F. Sunny, A. Mirza, M. Nikdast, and S. Pasricha, "CrossLight: A crosslayer optimized silicon photonic neural network accelerator," in *Proc. 58th* ACM/IEEE Design Autom. Conf. (DAC), Dec. 2021, pp. 1069–1074.
- [85] F. P. Sunny, A. Mirza, M. Nikdast, and S. Pasricha, "ROBIN: A robust optical binary neural network accelerator," ACM Trans. Embedded Comput. Syst., vol. 20, no. 5s, pp. 1–24, Oct. 2021.
- [86] F. Sunny, M. Nikdast, and S. Pasricha, "SONIC: A sparse neural network inference accelerator with silicon photonics for energy-efficient deep learning," 2021, arXiv:2109.04459.
- [87] F. Sunny, M. Nikdast, and S. Pasricha, "RecLight: A recurrent neural network accelerator with integrated silicon photonics," in *Proc. IEEE Comput. Soc. Annu. Symp. VLSI (ISVLSI)*, Jul. 2022, pp. 98–103.
- [88] S. Banerjee, C.-Y. Chen, J. Talukdar, S.-C. Hung, A. Chaudhuri, M. Nikdast, and K. Chakrabarty, "Towards functionally robust AI accelerators," in *Proc. IEEE Microelectron. Design Test Symp. (MDTS)*, May 2021, pp. 1–6.
- [89] S. Banerjee, M. Nikdast, and K. Chakrabarty, "Optimizing coherent integrated photonic neural networks under random uncertainties," in *Proc. Opt. Fiber Commun. Conf. (OFC)*, 2021, pp. 1–3.
- [90] S. Banerjee, M. Nikdast, and K. Chakrabarty, "Modeling silicon-photonic neural networks under uncertainties," in *Proc. Design, Autom. Test Eur. Conf. Exhib. (DATE)*, Feb. 2021, pp. 98–101.
- [91] S. Banerjee, M. Nikdast, and K. Chakrabarty, "Characterizing coherent integrated photonic neural networks under imperfections," *J. Lightw. Tech*nol., early access, Aug. 2, 2022, doi: 10.1109/JLT.2022.3193658.
- [92] C. D. Wright, P. Hosseini, and J. A. V. Diosdado, "Beyond von-Neumann computing with nanoscale phase-change memory devices," *Adv. Funct. Mater.*, vol. 23, no. 18, pp. 2248–2254, 2013.
- [93] M. Miscuglio, G. C. Adam, D. Kuzum, and V. J. Sorger, "Roadmap on material-function mapping for photonic-electronic hybrid neural networks," *APL Mater.*, vol. 7, no. 10, Oct. 2019, Art. no. 100903.

- [94] A. Sebastian, M. Le Gallo, and E. Eleftheriou, "Computational phasechange memory: Beyond von Neumann computing," J. Phys. D, Appl. Phys., vol. 52, no. 44, Oct. 2019, Art. no. 443002.
- [95] A. Sebastian, M. L. Gallo, R. Khaddam-Aljameh, and E. Eleftheriou, "Memory devices and applications for in-memory computing," *Nature Nanotechnol.*, vol. 15, no. 7, pp. 529–544, Mar. 2020.
- [96] F. P. Sunny, E. Taheri, M. Nikdast, and S. Pasricha, "A survey on silicon photonics for deep learning," ACM J. Emerg. Technol. Comput. Syst., vol. 17, no. 4, pp. 1–57, Oct. 2021.
- [97] A. Mirza, A. Shafiee, S. Banerjee, K. Chakrabarty, S. Pasricha, and M. Nikdast, "Characterization and optimization of integrated siliconphotonic neural networks under fabrication-process variations," *IEEE Trans. Nanotechnol.*, vol. 21, pp. 763–771, 2022.
- [98] T. F. de Lima, H. Peng, A. N. Tait, M. A. Nahmias, H. B. Miller, B. J. Shastri, and P. R. Prucnal, "Machine learning with neuromorphic photonics," *J. Lightw. Technol.*, vol. 37, no. 5, pp. 1515–1534, Mar. 1, 2019.
- [99] J.-M. Hung, C.-J. Jhang, P.-C. Wu, Y.-C. Chiu, and M.-F. Chang, "Challenges and trends of nonvolatile in-memory-computation circuits for AI edge devices," *IEEE Open J. Solid-State Circuits Soc.*, vol. 1, pp. 171–183, 2021.
- [100] C. Ríos, N. Youngblood, Z. Cheng, M. Le Gallo, W. H. P. Pernice, C. D. Wright, A. Sebastian, and H. Bhaskaran, "In-memory computing on a photonic platform," *Sci. Adv.*, vol. 5, no. 2, Feb. 2019, Art. no. eaau5759.
- [101] Z. Cheng, C. Ríos, W. H. P. Pernice, C. D. Wright, and H. Bhaskaran, "On-chip photonic synapse," *Sci. Adv.*, vol. 3, no. 9, Sep. 2017, Art. no. e1700160.
- [102] H. Zhang, B. Huang, Z. Zhang, C. Cheng, Z. Zhang, H. Zhang, R. Chen, and H. Chen, "On-chip photonic synapses based on slot-ridge waveguides with PCMs for in-memory computing," *IEEE Photon. J.*, vol. 13, no. 2, pp. 1–13, Apr. 2021.
- [103] I. Chakraborty, G. Saha, and K. Roy, "Photonic in-memory computing primitive for spiking neural networks using phase-change materials," *Phys. Rev. A, Gen. Phys.*, vol. 11, no. 1, Jan. 2019, Art. no. 014063.
- [104] P. Guo, N. Zhou, W. Hou, and L. Guo, "Starlight: A photonic neural network accelerator featuring a hybrid mode-wavelength division multiplexing and photonic nonvolatile memory," *Opt. Exp.*, vol. 30, no. 20, pp. 37051–37065, 2022.
- [105] V. Bangari, B. A. Marquez, H. Miller, A. N. Tait, M. A. Nahmias, T. F. de Lima, H.-T. Peng, P. R. Prucnal, and B. J. Shastri, "Digital electronics and analog photonics for convolutional neural networks (DEAP-CNNs)," *IEEE J. Sel. Topics Quantum Electron.*, vol. 26, no. 1, pp. 1–13, Jan. 2020.
- [106] Y. Wang, J. Ning, L. Lu, M. Bosman, and R. E. Simpson. (2021). *Multiphysics GCA Codes*. [Online]. Available: https://github. com/YunzhengWang/multiphysics_GCA
- [107] S. Senkader and C. D. Wright, "Models for phase-change of Ge₂Sb₂Te₅ in optical and electrical memory devices," *J. Appl. Phys.*, vol. 95, no. 2, pp. 504–511, Jan. 2004.
- [108] F. Tabatabaei, G. Boussinot, R. Spatschek, E. A. Brener, and M. Apel, "Phase field modeling of rapid crystallization in the phase-change material AIST," J. Appl. Phys., vol. 122, no. 4, Jul. 2017, Art. no. 045108.
- [109] A. Sebastian, M. Le Gallo, and D. Krebs, "Crystal growth within a phase change memory cell," *Nature Commun.*, vol. 5, no. 1, pp. 1–9, Jul. 2014.
- [110] P. Ashwin, B. S. V. Patnaik, and C. D. Wright, "Fast simulation of phase-change processes in chalcogenide alloys using a gillespie-type cellular automata approach," *J. Appl. Phys.*, vol. 104, no. 8, Oct. 2008, Art. no. 084901.
- [111] S. Yu and P.-Y. Chen, "Emerging memory technologies: Recent trends and prospects," *IEEE Solid State Circuits Mag.*, vol. 8, no. 2, pp. 43–56, Spring 2016.
- [112] S. Mittal and J. S. Vetter, "A survey of software techniques for using non-volatile memories for storage and main memory systems," *IEEE Trans. Parallel Distrib. Syst.*, vol. 27, no. 5, pp. 1537–1550, May 2016.
- [113] P. Fantini, S. Brazzelli, E. Cazzini, and A. Mani, "Band gap widening with time induced by structural relaxation in amorphous Ge₂Sb₂Te₅ films," *Appl. Phys. Lett.*, vol. 100, no. 1, Jan. 2012, Art. no. 013505.
- [114] J. Y. Raty, W. Zhang, J. Luckas, C. Chen, R. Mazzarello, C. Bichara, and M. Wuttig, "Aging mechanisms in amorphous phase-change materials," *Nature Commun.*, vol. 6, no. 1, pp. 1–8, Jun. 2015.
- [115] C. Li, C. Hu, J. Wang, X. Yu, Z. Yang, J. Liu, Y. Li, C. Bi, X. Zhou, and W. Zheng, "Understanding phase-change materials with unexpectedly low resistance drift for phase-change memories," *J. Mater. Chem. C*, vol. 6, no. 13, pp. 3387–3394, 2018.

- [116] M. Rütten, M. Kaes, A. Albert, M. Wuttig, and M. Salinga, "Relation between bandgap and resistance drift in amorphous phase change materials," *Sci. Rep.*, vol. 5, no. 1, pp. 1–11, Dec. 2015.
- [117] L. H. K. Duong, Z. Wang, M. Nikdast, J. Xu, P. Yang, Z. Wang, Z. Wang, R. K. V. Maeda, H. Li, X. Wang, S. Le Beux, and Y. Thonnart, "Coherent and incoherent crosstalk noise analyses in interchip/intrachip optical interconnection networks," *IEEE Trans. Very Large Scale Integr. (VLSI) Syst.*, vol. 24, no. 7, pp. 2475–2487, Jul. 2016.
- [118] M. Nikdast, J. Xu, L. H. K. Duong, X. Wu, X. Wang, Z. Wang, Z. Wang, P. Yang, Y. Ye, and Q. Hao, "Crosstalk noise in WDM-based optical networks-on-chip: A formal study and comparison," *IEEE Trans. Very Large Scale Integr. (VLSI) Syst.*, vol. 23, no. 11, pp. 2552–2565, Nov. 2014.
- [119] Y. Xie, M. Nikdast, J. Xu, X. Wu, W. Zhang, Y. Ye, X. Wang, Z. Wang, and W. Liu, "Formal worst-case analysis of crosstalk noise in mesh-based optical networks-on-chip," *IEEE Trans. Very Large Scale Integr. (VLSI) Syst.*, vol. 21, no. 10, pp. 1823–1836, Oct. 2013.
- [120] M. Nikdast, L. H. Duong, J. Xu, S. Le Beux, X. Wu, Z. Wang, P. Yang, and Y. Ye, "CLAP: A crosstalk and loss analysis platform for optical interconnects," in *Proc. IEEE/ACM Int. Symp. Networks Chip (NoCS)*, Sep. 2014, pp. 172–173.
- [121] M. Bahadori, S. Rumley, H. Jayatilleka, K. Murray, N. A. F. Jaeger, L. Chrostowski, S. Shekhar, and K. Bergman, "Crosstalk penalty in microring-based silicon photonic interconnect systems," *J. Lightw. Technol.*, vol. 34, no. 17, pp. 4043–4052, Sep. 1, 2016.
- [122] H. Wang, T. Guo, Y. Xue, S. Lv, D. Yao, Z. Zhou, S. Song, and Z. Song, "The phase change memory features high-temperature characteristic based on Ge-Sb-Se-Te alloys," *Mater. Lett.*, vol. 254, pp. 182–185, Nov. 2019.
- [123] S. Abdollahramezani, O. Hemmatyar, H. Taghinejad, A. Krasnok, Y. Kiarashinejad, M. Zandehshahvar, A. Alù, and A. Adibi, "Tunable nanophotonics enabled by chalcogenide phase-change materials," *Nanophotonics*, vol. 9, no. 5, pp. 1189–1241, 2020.
- [124] I. Boybat, B. Kersting, S. G. Sarwat, X. Timoneda, R. L. Bruce, M. BrightSky, M. L. Gallo, and A. Sebastian, "Temperature sensitivity of analog in-memory computing using phase-change memory," in *IEDM Tech. Dig.*, Dec. 2021, pp. 3–28.
- [125] A. Antolini, E. Franchi Scarselli, A. Gnudi, M. Carissimi, M. Pasotti, P. Romele, and R. Canegallo, "Characterization and programming algorithm of phase change memory cells for analog in-memory computing," *Materials*, vol. 14, no. 7, p. 1624, Mar. 2021.
- [126] A. Palmieri, A. Shafiee, M. G. C. Alasio, A. Tibaldi, G. Ghione, F. Bertazzi, M. Goano, and M. Vallone, "Enhanced dynamic properties of Ge-on-Si mode-evolution waveguide photodetectors," in *Proc. Int. Conf. Numer. Simul. Optoelectronic Devices (NUSOD)*, Sep. 2020, pp. 1–2.
- [127] C. Lian, C. Vagionas, T. Alexoudi, N. Pleros, N. Youngblood, and C. Ríos, "Photonic (computational) memories: Tunable nanophotonics for data storage and computing," *Nanophotonics*, vol. 11, no. 17, pp. 3823–3854, Aug. 2022.
- [128] A. Mirza, F. Sunny, P. Walsh, K. Hassan, S. Pasricha, and M. Nikdast, "Silicon photonic microring resonators: A comprehensive design-space exploration and optimization under fabrication-process variations," *IEEE Trans. Comput.-Aided Design Integr. Circuits Syst.*, vol. 41, no. 10, pp. 3359–3372, Oct. 2022.
- [129] M. Nikdast, G. Nicolescu, J. Trajkovic, and O. Liboiron-Ladouceur, "An analytical study of process variations in silicon photonic integrated circuits," in *Proc. Photon. North (PN)*, May 2016, pp. 1–2.
- [130] A. Shafiee, A. Mirza, F. Sunny, S. Banerjee, K. Chakrabarty, S. Pasricha, and M. Nikdast, "Inexact silicon photonics: From devices to applications," in *Proc. Photon. Switching Comput.* Washington, DC, USA: Optica Publishing Group, 2021, pp. 1–2. [Online]. Available: http://opg.optica.org/abstract.cfm?URI=PSC-2021-M3C.2
- [131] M. Nikdast, G. Nicolescu, J. Trajkovic, and O. Liboiron-Ladouceur, "Photonic integrated circuits: A study on process variations," in *Proc. Opt. Fiber Commun. Conf.* Washington, DC, USA: Optical Society of America, 2016, pp. 1–3.
- [132] P. Lazarenko, V. Kovalyuk, P. An, A. Prokhodtsov, A. Golikov, A. Sherchenkov, S. Kozyukhin, I. Fradkin, G. Chulkova, and G. Goltsman, "Size effect of the Ge₂Sb₂Te₅ cell atop the silicon nitride O-ring resonator on the attenuation coefficient," *APL Mater.*, vol. 9, no. 12, Dec. 2021, Art. no. 121104.
- [133] S. Raoux, R. M. Shelby, J. Jordan-Sweet, B. Munoz, M. Salinga, Y.-C. Chen, Y.-H. Shih, E.-K. Lai, and M.-H. Lee, "Phase change materials and their application to random access memory technology," *Microelectron. Eng.*, vol. 85, no. 12, pp. 2330–2333, Dec. 2008.

- [134] S. Raoux, J. L. Jordan-Sweet, and A. J. Kellock, "Crystallization properties of ultrathin phase change films," *J. Appl. Phys.*, vol. 103, no. 11, Jun. 2008, Art. no. 114310.
- [135] X. S. Miao, T. C. Chong, Y. M. Huang, K. G. Lim, P. K. Tan, and L. P. Shi, "Dependence of optical constants on film thickness of phasechange media," *Jpn. J. Appl. Phys.*, vol. 38, no. 3S, p. 1638, Mar. 1999.
- [136] W. Chong and Z. Koon, "Thickness dependent nano-crystallization in Ge₂Sb₂Te₅ films and its effect on devices," *Japanese J. Appl. Phys.*, vol. 46, no. 4S, p. 2211, 2007.
- [137] S. Raoux, G. W. Burr, M. J. Breitwisch, C. T. Rettner, Y. C. Chen, R. M. Shelby, M. Salinga, D. Krebs, S. H. Chen, H. L. Lung, and C. H. Lam, "Phase-change random access memory: A scalable technology," *IBM J. Res. Develop.*, vol. 52, nos. 4–5, pp. 465–479, 2008.
- [138] S. Mittal and A. I. Alsalibi, "A survey of techniques for improving security of non-volatile memories," *J. Hardw. Syst. Secur.*, vol. 2, no. 2, pp. 179–200, Jun. 2018.
- [139] T. Yu, X. Ma, E. Pastor, J. K. George, S. Wall, M. Miscuglio, R. E. Simpson, and V. J. Sorger, "All-chalcogenide programmable alloptical deep neural networks," 2021, arXiv:2102.10398.
- [140] M. Rudč, J. Pello, R. E. Simpson, J. Osmond, G. Roelkens, J. J. Van Der Tol, and V. Pruneri, "Optical switching at 1.55 μm in silicon racetrack resonators using phase change materials," *Appl. Phys. Lett.*, vol. 103, no. 14, Sep. 2013, Art. no. 141119.
- [141] C. Ríos, Q. Du, Y. Zhang, C.-C. Popescu, M. Y. Shalaginov, P. Miller, C. Roberts, M. Kang, K. A. Richardson, T. Gu, S. A. Vitale, and J. Hu, "Ultra-compact nonvolatile phase shifter based on electrically reprogrammable transparent phase change materials," 2021, arXiv:2105.06010.
- [142] C. Ríos, Q. Du, Y. Zhang, M. Shalaginov, P. Miller, P. Robinson, C. Roberts, M. Kang, K. A. Richardson, T. Gu, S. Vitale, and J. Hu, "Integrated nonvolatile phase-shifter based on electrically reconfigurable low-loss phase-change materials," in *Proc. Conf. Lasers Electro-Opt.*, 2021, pp. 1–15.
- [143] Q. Wilmart, H. El Dirani, N. Tyler, D. Fowler, S. Malhouitre, S. Garcia, M. Casale, S. Kerdiles, K. Hassan, C. Monat, X. Letartre, A. Kamel, M. Pu, K. Yvind, L. Oxenløwe, W. Rabaud, C. Sciancalepore, B. Szelag, and S. Olivier, "A versatile silicon-silicon nitride photonics platform for enhanced functionalities and applications," *Appl. Sci.*, vol. 9, no. 2, p. 255, Jan. 2019.
- [144] E. Gemo, J. Faneca, S. G.-C. Carrillo, A. Baldycheva, W. H. P. Pernice, H. Bhaskaran, and C. D. Wright, "A plasmonically enhanced route to faster and more energy-efficient phase-change integrated photonic memory and computing devices," *J. Appl. Phys.*, vol. 129, no. 11, Mar. 2021, Art. no. 110902.
- [145] M. Miscuglio and V. J. Sorger, "Photonic tensor cores for machine learning," *Appl. Phys. Rev.*, vol. 7, no. 3, Sep. 2020, Art. no. 031404.
- [146] J. Feldmann, N. Youngblood, M. Karpov, H. Gehring, X. Li, M. Stappers, M. Le Gallo, X. Fu, A. Lukashchuk, A. S. Raja, J. Liu, C. D. Wright, A. Sebastian, T. J. Kippenberg, W. H. P. Pernice, and H. Bhaskaran, "Parallel convolutional processing using an integrated photonic tensor core," *Nature*, vol. 589, no. 7840, pp. 52–58, Jan. 2021.
- [147] F. Brückerhoff-Plückelmann, J. Feldmann, C. D. Wright, H. Bhaskaran, and W. H. P. Pernice, "Chalcogenide phase-change devices for neuromorphic photonic computing," *J. Appl. Phys.*, vol. 129, no. 15, Apr. 2021, Art. no. 151103.
- [148] F. Brückerhoff-Plückelmann, J. Feldmann, H. Gehring, W. Zhou, C. D. Wright, H. Bhaskaran, and W. Pernice, "Broadband photonic tensor core with integrated ultra-low crosstalk wavelength multiplexers," *Nanophotonics*, vol. 11, no. 17, pp. 4063–4072, Aug. 2022.
- [149] I. Giannopoulos, A. Sebastian, M. Le Gallo, V. P. Jonnalagadda, M. Sousa, M. N. Boon, and E. Eleftheriou, "8-bit precision in-memory multiplication with projected phase-change memory," in *IEDM Tech. Dig.*, Dec. 2018, pp. 7–27.
- [150] X. Yang, M. S. Nisar, W. Yuan, F. Zheng, L. Lu, J. Chen, and L. Zhou, "Phase change material enabled 2 × 2 silicon nonvolatile optical switch," *Opt. Lett.*, vol. 46, no. 17, pp. 4224–4227, 2021.
- [151] M. Stegmaier, C. Ríos, H. Bhaskaran, C. D. Wright, and W. H. P. Pernice, "Nonvolatile all-optical 1 × 2 switch for chipscale photonic networks," *Adv. Opt. Mater.*, vol. 5, no. 1, Jan. 2017, Art. no. 1600346.
- [152] M. S. Arjunan, S. Durai, and A. Manivannan, "Multilevel switching in phase-change photonic memory devices," *Phys. Status Solidi (RRL) Rapid Res. Lett.*, vol. 15, no. 11, Nov. 2021, Art. no. 2100291.
- [153] M. Zhang, C. Wang, P. Kharel, D. Zhu, and M. Lončar, "Integrated lithium niobate electro-optic modulators: When performance meets scalability," *Optica*, vol. 8, no. 5, pp. 652–667, 2021.

- [154] W. Bogaerts, P. De Heyn, T. Van Vaerenbergh, K. De Vos, S. Kumar Selvaraja, T. Claes, P. Dumon, P. Bienstman, D. Van Thourhout, and R. Baets, "Silicon microring resonators," *Laser Photon. Rev.*, vol. 6, no. 1, pp. 47–73, Jan. 2012.
- [155] E. Taheri, S. Pasricha, and M. Nikdast, "ReSiPI: A reconfigurable siliconphotonic 2.5 D chiplet network with PCMs for energy-efficient interposer communication," in *Proc. 41st IEEE/ACM Int. Conf. Comput.-Aided Design*, Oct. 2022, pp. 1–9.
- [156] Q. Zhang, H. Yu, M. Barbiero, B. Wang, and M. Gu, "Artificial neural networks enabled by nanophotonics," *Light, Sci. Appl.*, vol. 8, no. 1, pp. 1–14, May 2019.
- [157] S. V. Reddy Chittamuru, I. G. Thakkar, and S. Pasricha, "Analyzing voltage bias and temperature induced aging effects in photonic interconnects for manycore computing," in *Proc. ACM/IEEE Int. Workshop Syst. Level Interconnect Predict. (SLIP)*, Jun. 2017, pp. 1–8.



AMIN SHAFIEE (Graduate Student Member, IEEE) received the B.Sc. degree in electronics engineering from Shiraz University, Iran, in 2018, and the M.S. degree in integrated electronics and optoelectronics from the Polytechnique University of Turin (Politecnico di Torino), Italy, in 2020. He is currently a Graduate Research Assistant at Colorado State University. His research interests include silicon photonics, photonic devices, and photodetectors.



SUDEEP PASRICHA (Senior Member, IEEE) received the B.E. degree in electronics and communications from the Delhi Institute of Technology, India, and the M.S. and Ph.D. degrees in computer science from the University of California at Irvine, Irvine, CA, USA. He is currently a Professor and the Chair of computer engineering at Colorado State University, where he is also a Professor of computer science and systems engineering. His research interests include design of innovative

software algorithms, hardware architectures, and hardware–software codesign techniques for energy-efficient, fault-tolerant, real-time, and secure computing. He is an ACM Distinguished Member. He has received 16 best paper awards and nominations at various IEEE and ACM conferences and several other awards for research excellence.



MAHDI NIKDAST (Senior Member, IEEE) received the Ph.D. degree in electronic and computer engineering from The Hong Kong University of Science and Technology (HKUST), Hong Kong, in 2014. From 2014 to 2017, he was a Postdoctoral Fellow at McGill University and Polytechnique Montreal, Canada. He is currently an Assistant Professor and an Endowed Rockwell-Anderson Professor at the Department of Electrical and Computer Engineering, Colorado

State University (CSU), Fort Collins, CO, USA. He is also the Director of the Electronic-PhotoniC System Design (ECSyD) Laboratory, CSU. His primary research interests include design and development of photonic systems-on-chip and next-generation data-communication, computing, and sensing systems employing emerging technologies, while emphasizing energy efficiency and robustness. He and his students have received multiple best paper awards for their work in the area of integrated photonics and design for manufacturability.